Assessing energy flexible operation of a residential Zero Emission Building (casaZero)

Master's thesis in Energy and Environmental Engineering Supervisor: Vojislav Novakovic Co-supervisor: Karen Byskov Lindberg June 2022

Master's thesis

Norwegian University of Science and Technology Faculty of Engineering Department of Energy and Process Engineering



Åshild Ryen Vespestad

Assessing energy flexible operation of a residential Zero Emission Building (casaZero)

Master's thesis in Energy and Environmental Engineering Supervisor: Vojislav Novakovic Co-supervisor: Karen Byskov Lindberg June 2022

Norwegian University of Science and Technology Faculty of Engineering Department of Energy and Process Engineering



Norwegian University of Science and Technology

EPT-M-2022



PROJECT WORK

for

student Åshild Ryen Vespestad

Spring 2022

l

Vurdering av energifleksibel drift av et nullutslippshus (casaZero)

Background and objective

Moving towards zero emission buildings, energy can be managed in a flexible way, to achieve e.g., reduced power peaks, reduced energy use, reduced CO₂-emissions, increased self-consumption of locally produced energy, or provide flexibility services to distribution system operators (DSOs).

The aim of this assignment is to investigate the impact of intelligent energy management in households, using data gathered from a real zero emission building situated in Oslo (casaZero). casaZero has roof-mounted PV, a ground-source heat pump, waterborne floor heating and a ventilation system. The goal is to establish a model that can simulate the house, and use this model to investigate different operational strategies and how these will affect the performance of the building regarding e.g. peak load, energy costs and self-consumption of PV, as well as thermal comfort of the residents and other indoor environmental parameters.

The following tasks are to be considered:

- 1. Review previous work on optimal and/or flexible operation of Zero Emission Buildings / Zero Energy Buildings.
- 2. Propose and develop a suitable simulation model of casaZero (using e.g. Modelica or IDA-ICE) that could be used for assessing the energy flexibility potential of the case building (casaZero). Preferably utilising some of the energy measurements gathered in the project thesis.
- 3. Investigate how different operational strategies of the ground source heat pump may impact the performance of the building.
- 4. Make a draft proposal (6-8 pages) for a scientific paper based on the main results of the work performed in the master thesis.
- 5. Make proposal for further work on the same topic.

-- " --

The master work comprises 30 ECTS credits.

Address N-7034 Trondheim Norway Location K. Hejes vei 1b Tel.+47 73 59 38 60Fax+47 73 59 35 80Org. no.NO 974 767 880

The work shall be edited as a scientific report, including a table of contents, a summary in Norwegian, conclusion, an index of literature etc. When writing the report, the candidate must emphasise a clearly arranged and well-written text. To facilitate the reading of the report, it is important that references for corresponding text, tables and figures are clearly stated both places. By the evaluation of the work the following will be greatly emphasised: The results should be thoroughly treated, presented in clearly arranged tables and/or graphics and discussed in detail.

The candidate is responsible for keeping contact with the subject teacher and teaching supervisors.

Risk assessment of the candidate's work shall be carried out according to the department's procedures. The risk assessment must be documented and included as part of the final report. Events related to the candidate's work adversely affecting the health, safety or security, must be documented and included as part of the final report. If the documentation on risk assessment represents a large number of pages, the full version is to be submitted electronically to the supervisor and an excerpt is included in the report.

According to "Utfyllende regler til studieforskriften for teknologistudiet/sivilingeniørstudiet ved NTNU" § 20, the Department of Energy and Process Engineering reserves all rights to use the results and data for lectures, research and future publications.

Submission deadline: June 15th, 2022

Work to be done in lab (Water power lab, Fluids engineering lab, Thermal engineering lab) Field work

Department for Energy and Process Engineering, January 2022

an tavatur

Vojislav Novakovic Supervisor

Co-Supervisor(s): Senior Research Scientist Karen Byskov Lindberg, SINTEF Community, e-mail: karen.lindberg@sintef.no

Preface

This Master's thesis was written at the Department of Energy and Process Engineering at the Norwegian University of Science and Technology, NTNU, in Trondheim. The work was conducted during the spring of 2022. The supervisor for this project was Vojislav Novakovic, and Karen Byskov Lindberg was the co-suprevisor.

The author would like to express gratitude for all the help and guidance provided by the supervisor Vojislav Novakovic. Furthermore, acknowledgement should be given to the co-supervisor Karen Byskov Lindberg for her great help and patience during the work of this thesis as well as her willingness to share data from casaZero. Natasa Nord should also be thanked for her help in providing licence and guidance for the work performed in IDA ICE. Last the author would like to thank NTNU for providing the opportunity to complete this master 's degree.

Trondheim, June 2022.

Åshild Ryen Vespestad

Abstract

With an increasing amount of new and retrofitted buildings qualifying as ZEBs, the interaction between buildings and the grid has increased. Buildings that both produces and consumes electricity, prosumers, are likely to both import and export electricity to the grid. Hence, the need to utilized smart controlling of the electricity consumption in buildings is reinforced, as an attempt to reduce the stress on the grid. This study is a contribution to increasing the knowledge on flexible operation of ZEBs. The case building in this research was a residential ZEB located in Norway. The building was equipped with a GSHP, a balanced ventilation system, PV panels and water borne floor heating. A simulation model of the case building was established in IDA ICE. By implementing 65 different operational schedules to the GSHP the impact on the performance of the building was analysed, regarding self-consumption, peak load and cost. It was found that the self-consumption could be increased by 13.7%. The summer season was found to be the most influential season considering further increasing of the self-consumption. By introducing an additional load simulating the battery of an EV, the self-consumption was further increased by 53.3%, compared to the case without a GSHP schedule. The maximum reduction for peak load and cost found in this research were 50%and 9.7%, respectively. When equally weighting the three KPIs, the results showed that a GSHP schedule with a constant value of 0.5 gave the best overall result. A relation between the different KPIs were discovered, making it challenging to decrease the cost while increasing the self-consumption.

Sammendrag

Med en økende andel nyoppsatte og oppussede bygg som tilfredsstiller dagens krav for nullutslippshus har også interaksjonen mellom bygninger og kraftnettet økt. Bygg som både produserer og konsumerer elektrisitet, prosumers, er ofte både importører og eksportører av elektrisitet til nettet. Av den grunn har behovet for smart styring av laster i bygg økt i et forsøk på å senke belastningen på nettet. Dette studiet er et bidrag til å øke kunnskapen om fleksibel drift av nullutslippshus. Bygget som ble studert i denne masteroppgaven er et nullutslipps bolighus i Norge. Bygget var utrustet med en bergvarmepumpe, balansert ventilasjon, solcellepaneler på taket og vannbåren gulvvarme. En simuleringsmodell av bygget ble laget i IDA ICE. Ved å implementere 65 ulike timeplaner for driften av varmepumpen i bolighuset ble påvirkningen på byggets energiprestasjon studert, med tanke på selv-konsumering, topplast og kostnad. Fra resultatene i oppgaven ble det funnet at selv-konsumeringen kunne økes med 13.7%. Sommeren var den årstiden som hadde størst potensialet for videre økning av selv-konsumeringen. Ved å implementere en last som representerte batteriet til en elbil ble selv-konsumeringen økt med 53.3%sammenlignet med modellen uten timeplan for styring av varmepumpa. Den største reduksjonen i topplast og kostnad var henholdsvis 50% og 9.7%. Når hver av de tre indikatorene for nøkkelytelse ble vektlagt likt viste resultatene at varmepumpetimeplanen med konstant verdi på 0.5 ga de beste resultatene sammenlagt. Det ble oppdaget en sammenheng mellom de ulike indikatorne for nøkkelytelse, noe som gjorde det vanskelig å redusere kostnadene samtidig som selv-konsumeringen skulle økes.

Abbreviations

- AHU Air Handling Unit
- $BIM\,$ Building Information Modeling
- BPS Building Performance Simulation
- $CHP\,$ Combined Heat and Power
- $DHW\,$ Domestic Hot Water
- $DSM\,$ Demand Side Management
- EV Electric Vehicle
- $GHG\,$ Green House Gas
- $GSHP\,$ Ground Source Heat Pump
- $HEMS\,$ Home Energy Management System
- *HP* Heat Pump
- HVAC Heating, Ventilation and Air Conditioning
- IDA ICE IDA Indoor Climate and Energy
- IFC Industry Foundation Classes
- KPI Key Performance Indicator
- $nZEB\,$ nearly Zero Emission Building
- PPD Percentage of People Dissatisfied
- PV Photo voltaic
- TES Thermal Energy Storage
- $ZEB\,$ Zero Emission Building
- ZEN Zero Emission Neighborhood

Contents

1	Intr	oduction	1
	1.1	Motivation	1
	1.2	Scope	2
	1.3	Limitation	2
	1.4	Thesis overview	3
2	The	ory	4
	2.1	ZEB definition	4
	2.2	Defining energy flexibility of buildings	6
	2.3	Demand Side Management	7
	2.4	Optimal and flexible operation of ZEB	8
		2.4.1 Optimizing cost	9
		2.4.2 Optimizing self-consumption	0
		2.4.3 Optimizing peak load	2
		2.4.4 Optimizing energy use	2
3	Met	hodology 1	4
Ŭ	3.1	IDA ICE	4
	3.2	CasaZero	6
		3.2.1 Building envelope	9
		3.2.2 Climate and location	9
		3.2.3 Internal gains and human behaviour	0
		3.2.4 Ventilation and AHU	1
		3.2.5 The plant	1
		3.2.6 Electricity prices	3
	3.3	Key Performance Indicators	4
		3.3.1 Self-consumption	4
		3.3.2 Peak load	5
		3.3.3 Cost	5
	3.4	Assessing the energy flexibility potential	5
4	Cas	e study 2	7
	4.1	Base case	7
	4.2	GSHP schedules	7
	4.3	Case 1, Case 2 and Case 3	7
5	Res	ults and Analysis 2	9
Ŭ	5.1	Base case 2	9
	5.2	Identifying Case 1. Case 2 and Case 3	1
	5.3	Case 1	2
	5.4	Case 2	5
	5.5	Case 3	9

	5.6 Optimal overall model	41
6	Discussion 6.0.1 Schedule for DHW tapping	43 43
7	Conclusion	45
8	Further work	46
A	Appendix 1	52
в	Appendix 2	53
С	Appendix 3	58
D	Appendix 4	60
\mathbf{E}	Appendix 5	62
F	Appendix 6	64
G	Appendix 7	66

List of Figures

1	The three basic principles for ZEBs	5
2	Energy balance for a ZEB-COMPLETE.	6
3	Categorized ways to achieve energy flexibility in a building	7
4	Demonstration of peak shaving, load shifting and valley filling	8
5	Example of how the load profile for a 24h period might not match the	
	electricity production in a residential building.	11
6	An excerpt of the delivered energy report given when running simulations	
	in IDA ICE	15
7	The chart flow for a process in IDA ICE, aiming to maximize the self-	
	consumption. Black lines represent processes that are only done once, while	
	the red lines represent repetitive processes that are done several times dur-	
	ing an optimization.	16
8	The actual floor plan for the 1. floor in casaZero drawn by Snøhetta	17
9	The floor plan for all three levels in casaZero made in IDA ICE	18
10	The orientation of casaZero.	18
11	The ambient air temperatures used for simulation in IDA ICE	20
12	The schedules for lighting and equipment in residential buildings based on	
	the schedules found in SN/TS 3031:2016	21
13	Part of the plant established in IDA ICE for casaZero	22
14	The technical equipment of casaZero including the electricity and heat flow	
	between the equipment	23
15	The electricity cost curve for December 2021 implemented in the model	23
16	Overview of the methodology used in this thesis	26
17	GSHP schedule for S1	27
18	Load profile for the 72 hour periods with highest electricity consumption	
	for winter and spring season for Base case.	30
19	Load profile for the 72 hour periods with highest electricity consumption	
	for summer and autumn season for Base case	30
20	Result from the first 20 schedules implemented in the simulation model.	
	The three best results considering each KPI is marked in green, yellow and	
	blue	31
21	The three schedules with highest score of the 20 first schedules. S13 gave	
	the lowest cost, S16 gave the highest self-consumption, while S7 gave the	
	lowest imported electricity peak	32
22	An overview of the main results from sensitivity analysis of Case 1	33
23	GSHP schedule for S16(3). \ldots	34
24	Electricity production vs electricity consumption for four selected days dur-	
	ing the one year simulation period with $S16(3)$	34
25	An overview of the main results from the sensitivity analysis of Case 2. $\ .$.	36
26	GSHP schedule for $S9(2)$	36
27	Monthly peak load for Base case, S7, S9, S7(1) and S9(2). \ldots	37

28	Load profile for the 72 hour periods with highest electricity consumption	
	for winter and spring season for $S9(2)$	38
29	Load profile for the 72 hour periods with highest electricity consumption	
	for summer and autumn season for $S9(2)$	38
30	An overview of the main results from the sensitivity analysis of Case 3. $\ .$.	39
31	Imported electricity for Base case, S12, S2(3) and S20(1). \ldots \ldots \ldots	40
32	GSHP schedules for S12, S2(3) and S20(1). \ldots	40
33	Monthly peak load for some selected schedules during sensitivity analysis	
	of Case 3	41
34	GSHP schedule for S7(1). \ldots \ldots \ldots \ldots \ldots \ldots \ldots	41
35	Load profile for the 72 hour periods with highest electricity consumption	
	for winter and summer season for $S7(1)$	42
36	The solar radiation of a 8 day period in January.	44

List of Tables

1	Overview of the physical parameters of the building	19
2	The self-consumption for $S16(3)$ and Base case for each season and for a	
	hole year	35
3	Main results from Base case and BCs	43
4	KPI values for the best schedule in each case and the overall best schedule	
	compared to the results from Base case	45

1 Introduction

This section introduces the motivation and the scope of this thesis as well as some limitations. Last, an overview of the content of this thesis is given.

1.1 Motivation

About one third of the global energy use and 40% of the total CO2 emission world wide today is caused by the building sector, during the constructional and operational phase [1]. To lower the CO2 emission in Europe the European Union has decided to reduce the CO2 emission associated with the building sector by 80-95%, compared to the 1990 level by 2050 [2]. To reach this goal Zero Emission Buildings (ZEBs) are an important part of the strategy. In 2010 the European Union stated that by the end of 2020 all new buildings should be nearly Zero Emission Buildings (nZEBs) [3].

As all buildings require a minimum amount of electricity, ZEBs depend on on-site renewable electricity production to fulfill the ZEB requirement. The renewable electricity sources available are often fluctuating, leaving the building dependent on electricity from the grid in the hours when the local electricity sources are not supplying sufficient amount of electricity to the building [4]. When the local electricity sources produce more electricity than consumed by the building, excess electricity can be supplied to the grid. This way of buying and selling electricity to the grid characterizes the building as a prosumer [5]. With an increasing number of prosumers intermittently supplying electricity to the grid, the grid companies are facing challenges handling the unpredictability in the grid. Both considering hourly mismatch and seasonal mismatch [6].This calls for an increased utilization of energy flexibility measures in ZEBs to increase the self-consumption of electricity in the building.

The energy flexibility measures are divided into three categories, load shifting measures, peak shaving measures and valley filling measures. The first measure aims to shift the load from hours with high electricity consumption to hours with lower electricity consumption. The second measure reduces the peak load, and the third measure aims to increase the consumption in periods with low consumption to flatten the curve [7]. By combining these measures the operation of a ZEB can be managed in an optimal and flexible way. One way of optimizing the operation of a ZEB is by parametric study. In this method all parameters except one are fixed. The non-fixed parameter is varied to find the optimal solution to the problem [8]. The more complex the building is, the more time consuming the optimization process becomes. To optimize the load profile in a more time effective way simulation based methods are needed. These can be used, for instance, in optimization of a building [9].

Since the term ZEB was introduced in 2006 [10], several research papers have been written focusing on optimizing ZEBs. New and improved technology has emerged, enabling communication between technical equipment in the building. This increases the possibility for operating buildings in an optimal way, considering cost, CO2 emission, indoor environment etc. Despite this, to the authors knowledge, the main research area on ZEB has been optimizing the design rather than the operation. Research like [6] and [11] are focusing on cost optimization of the design of a ZEB, while [12] is focusing on optimizing the design to minimize green house gas (GHG) emission. In a research done in a climate with cold winters and hot summers, an optimization of design indicated almost 35% reduction in electricity consumption of a residential building [13]. To find out how optimal operation of a ZEB can effect the energy use, cost, CO2 emission etc. more research on optimization of the operation of ZEBs is coveted.

1.2 Scope

To contribute to more research on optimal and flexible operation of ZEBs, this thesis aims to establish a model to simulate a ZEB in order to investigate the energy flexibility potential of the building. The main objective of this research is to assess energy flexible operation of a residential ZEB, named casaZero. The building is located in Norway and it is equipped with photo voltaic (PV) panels, a ground source heat pump (GSHP), a balanced ventilation system and a water borne heating system. By the use of simulation software, different operational strategies are to be investigated in regard to peak load, cost and self-consumption.

To perform this analysis the task is divided into five main parts:

- Perform a literature review on previous research on optimal or flexible operation of ZEBs.
- Establish a model of the case building to be used in assessment of the energy flexibility potential of casaZero.
- Apply different operational strategies to the GSHP to find the influence on the performance of the building.
- Make a proposal for further work on the same topic.
- Write a draft for a scientific paper based on an excerpt of the results found in this master thesis.

1.3 Limitation

Amongst the limitations found in this research are the many assumptions necessary to establish a simulation model in IDA ICE. Several schedules were implemented for cost, tapping of domestic hot water (DHW), occupants, temperature etc. To produce more accurate results real measured data should be used. The household possesses an electric vehicle (EV) that is considered a shiftable load. Only one simulation was run with a load corresponding to the battery of an EV. To find the energy flexibility potential of casaZero with the EV, more simulations should be done including the EV. Further, IDA ICE does not offer the possibility to establish schedules with seasonal differences. Applying different schedules for different seasons would provide better results, due to the large variation in electricity consumption for the different seasons.

1.4 Thesis overview

The content of this thesis is divided into 8 chapters. Chapter 2 includes background information and a literature review on optimal and flexible operation of ZEBs. Chapter 3 describes the utilized methodology, including the simulation software, a detailed description of the case building and the key performance indicators (KPIs) utilized to analyse the results. Chapter 4 describes the cases studied in this thesis. Chapter 5 includes the results. Chapter 6 discusses some limitations of this research. Chapter 7 is the conclusion of the work and Chapter 8 includes suggestions for further work on the topic.

2 Theory

This section describes the background information for this master thesis. It includes definitions of ZEBs, energy flexibility and demand side management (DSM). The last part of this chapter is a literature review of previous research on optimal and flexible operation of ZEBs.

2.1 ZEB definition

Before discussing the operation of a ZEB, a definition of the ZEB-concept must be established. A ZEB is usually referred to as an energy efficient building with on-site renewable energy generation sized to generate at least the amount of energy used by the building [10]. This can be achieved by designing a house with a minimum need of energy combined with utilization of local renewable energy sources [14]. The exact definition and requirement for a ZEB differs from country to country. One such difference is the minimum requirement for utilization of renewable energy. While the US does not have any requirements, the EU has a minimum number of 20% of the total energy use of the building [15]. The targeted energy use does also vary for these two regions. In addition to regulations differing from countries, the local climate condition is an important factor when deciding the design and technical solutions of a ZEB. Solutions to reduce the cooling demand is important in a hot climate [16], while in a cold climate, solutions to reduce the heating demand is vital to meet the local requirements for a ZEB [11].

The three important principles when designing and operating a ZEB is shown in Figure 1. The first principle is to avoid unnecessary energy use. This includes constructing the building envelope to avoid thermal bridges and to minimize heat loss through external surfaces. The second principle is to reduce energy use by, for instance, increasing the energy efficiency of technical installations in the building. The last, and most vital principle to qualify as a ZEB is replacing the energy used in the building by on-site renewable energy generation.



Figure 1: The three basic principles for ZEBs.

The ZEB research centre [17] has organized ZEBs into different sub categories depending on the ambition levels for the zero emission balance of the building.

- ZEB-O: The building generates as much renewable energy as required to compensate for GHG emission during operation of the building.
- ZEB-O÷EQ: The building generates as much renewable energy as required to compensate for GHG emission during operation of the building, minus the energy used by equipment.
- ZEB-OM: The building generates as much renewable energy as required to compensate for GHG emission during operation of the building and during production of the used materials.
- ZEB-COM: The building generates as much renewable energy as required to compensate for GHG emission during operation, during production of the used materials and during the construction of the building.
- ZEB-COMPLETE: The building generates as much renewable energy as required to compensate for GHG emission for the entire life cycle of the building.

Figure 2 shows the energy balance in a ZEB-COMPLETE with a renewable energy generation balancing the energy use for all four phases of the building life cycle.



Figure 2: Energy balance for a ZEB-COMPLETE.

2.2 Defining energy flexibility of buildings

With increased focus on ZEBs, the utilization of on-site fluctuating renewable energy sources is rising. This has led to a growing interest in utilization of the energy flexibility potential of buildings. However, it is not easy to define the energy flexibility that a building can provide [18]. This is due to the large amount of factors effecting the energy flexibility. The Cambridge dictionary definition of flexibility is "the ability to change or be changed easily according to the situation". Hence, the definition of energy flexibility of buildings would also include the ability to change according to the situation. The International Energy Agency defines energy flexibility in a building as "the capacity of a building to manage its demand and generation according to local climate conditions, user needs and grid requirements" [19].

Figure 3 shows an overview of typical factors contributing to the potential of energy flexibility in a building. Building mass, thermal storage and batteries are all offering energy flexibility potential by storing energy, so that energy peaks can be shifted. Generation provides a potential of self-consumption to the building that might ease the pressure on the grid. Fuel shifting is based on the concept of utilizing several fuels to be able to switch the energy source in the building depending on, for instance, the price of the different fuels. A combination of a heat pump (HP) and a gas boiler to satisfy the heating demand, is an example of increasing the potential of energy flexibility in a building by fuel shifting. The last category is the flexibility provided by connecting the building to a grid.



Figure 3: Categorized ways to achieve energy flexibility in a building.

Though the building might include several of the components displayed in Figure 3, the ability to utilize the energy flexibility depends on implementation of control strategies [18]. Such strategies might vary in complexity from a simple schedule controlling on and off of a ventilation system, to complicated model based controlling strategies where load forecasting, price estimation and occupancy behaviour are some of the factors influencing the system.

2.3 Demand Side Management

An example of utilization of energy flexibility in a building by implementing control strategies is the demand side management (DSM). DSM aims to balance the demand and supply in a smart energy system. To optimize building operation with respect to GHG emission, minimizing the gap between energy generation and energy use is of importance according to Farrokhi et al. [20]. To enable DSM in a building a communication infrastructure between the equipment of the building is important. Through communication, the different equipment and technical installations in the building can interact towards optimal operation. DSM can be utilized to enable energy flexibility measures in a buildings, such as valley filling and peak shaving. All three measures are displayed in Figure 4. DSM can perform instantaneously optimization of building operation by being an integrated part of the system in a building. This requires a load prediction model to estimate the future load for the building.



Figure 4: Demonstration of peak shaving, load shifting and valley filling.

Night time heating is an example of peak shaving implemented in some office buildings [21]. This allows heating to be performed during the night or early in the morning, before employees arrive. The office can be provided with comfortable temperatures and at the same time reduce the peak load during the day by utilization of thermal energy storage (TES) in the building. This example shows how technical installations of the building can be controlled to contribute to a more optimal operation of a building.

2.4 Optimal and flexible operation of ZEB

There are several reasons for utilizing the energy flexibility of a building through optimization of the building operation. Researchers have pointed out reduction of stress on the electricity grid [22], cost reduction and CO2 emission reduction [23] as some important factors. In this subsection, previous literature on the topic is used to conclude how energy flexibility of buildings can be evaluated.

Optimization of a process can be both single objective optimization and multi objective optimization, depending on the amount of objectives for the optimization process [24]. The goal for both single and multi objective optimization of building operation is to find the solution that optimizes the objective function based on a set of given constraints. In building operation the objective function can be minimization of cost, energy demand, GHG emission, peak load etc. Examples of constraints are the energy balance, indoor air temperature, air quality and other factors influencing the comfort of the occupants of the building. It has been shown that optimizing one parameter might degrade the results of other parameters [25]. When performing optimization of building operation, multi objective optimization is more convenient as it considers multiple objectives and constraints. Handling multi objective optimization problems requires a large amount of various parameter combinations. To optimize such problems in a time efficient way, optimization tools can be utilized. One such tool is the optimization software GenOpt. This software is developed to minimize a cost function. By iterative calling of building energy modeling (BEM) softwares like IDA ICE, TRNSYS or EnergyPlus, GenOpt evaluates the objective function. GenOpt has been used in research like [9] and [26]. Another software used for optimization is FICO Xpress Mosel. This optimization language can be used to develop models to solve comprehensive optimization problems. In a research done by Lindberg et al. [6] this optimization language was utilized to evaluate the cost-optimal solution for an energy system as well as the hourly operation of a ZEB.

Another example of combining a dynamic energy simulation tool and an optimization tool like GenOpt can be found in a research performed by Ferrara et al. [23], in 2019. The method was used to cost optimize the energy efficient measures of a nearly zero emission multi-family building, located in northern Italy. In the research, the energy simulation program TRNSYS was used to establish the objective function, and a suitable optimization algorithm was chosen in GenOpt. This software can be used to evaluate, not only the optimal solution, but also neighbouring values. Hence, alternative solutions to the optimal value can be considered. In the mentioned research this resulted in findings that showed the possibility of decreasing the energy demand per square meter by slightly increase the total cost.

The review of previous research on flexible and optimal operation of ZEBs found that the most common objectives are minimizing the cost and maximizing self-consumption. Furthermore, it was found that most research uses multi objective optimization, focusing on several objectives at a time. The following sections describes the most common optimization objectives in ZEB operation.

2.4.1 Optimizing cost

An important factor when designing and operating a ZEB is the cost. Since economics is a governing factor in a lot of decisions made both in private life and on a national level, reducing the cost of building operation will always be of interest.

When minimizing the operational cost of a ZEB, it is important to include the thermal comfort of the occupants as a restriction. This is in addition to the restrictions given by the definition of a ZEB. This was done in a research performed by Zhang et al. [27]. The aim of their research was to find the optimal operation, considering both minimizing electricity cost and maximizing comfort for the occupants. By making use of a home energy management system (HEMS) the imported, exported and consumed electricity by the different installations were monitored. The HEMS was equipped with a controller that communicated with the components through a home area network. The HEMS also ac-

cessed weather prediction, that enabled it to make estimations for both PV production and heat load for the building. During the process of establishing the optimization algorithm the research team utilized weighting methods to merge the two optimization objects into a single optimization object. This resulted in an algorithm implemented to the HEMS that optimized the cost and comfort properties for the next 24 hours.

Lindberg et al. [6] investigated the energy systems in a ZEB. They found that the variety in ZEB definitions might influence the cost optimization of a building. In a cost optimization research utilizing three different grades of ZEB, "no"-ZEB, "nearly"-ZEB and "Strictly"-ZEB, the cost optimal energy system and hourly operation was studied. They differed regarding to their level of ZEB, hence, the level of gap between generation and use of energy. It was found that the optimal operation of a ZEB might differ from country to country due to different requirements regarding energy use and generation in ZEBs.

When considering cost of operation of a building, the concept of penalty cost due to emission should be mentioned. This was the focus in a research performed by Thorvaldsen et al. [28]. The aim was to cost-optimize the operation of a ZEB while considering costs due to emission compensation. It was found that with a low penalty cost for emission, the operation was regulated by the electricity price. This caused load shifting to hours of the day when the electricity price was low. With high penalty cost the operation of the building was changed, increasing the load during hours with low emission intensity. Cost optimal operation of a ZEB will therefore vary depending on penalty cost of CO2 emission.

2.4.2 Optimizing self-consumption

While self-sufficiency is the amount of the energy demand that is covered by the on-site generation, self-consumption is the share of the produced electricity that is consumed locally [29]. To ease the pressure on the grid, the self-consumption should be as high as possible.

Though a building might have a high rate of self-sufficiency, the self-consumption can still be low without proper energy storage or load control strategies. Figure 5 shows a fictive curve of the load profile and the electricity production in a residential building for a 24 hour period. The need for load matching is clearly vital when optimizing the selfconsumption of a ZEB. Lopes et al. [30] divides the approaches found in research on load matching into three, energy storage, DSM and a combination of the two. Most of the researches including energy storage consider TES or electrical batteries, but some research analyzes the ability of hydro storage to improve the load matching [31]. The following sections confirms that by utilizing energy storage and control strategies, it is possible to improve the self-consumption of a ZEB.



Figure 5: Example of how the load profile for a 24h period might not match the electricity production in a residential building.

Dar et al. [25] did a research on improving the energy flexibility of a ZEB based on HP control. The chosen energy storage for the research was the TES, and the research included investigation of three control strategies. It showed that the self-consumption could be increased by 40% with proper control of the HP. In the case of the research a seasonal storage could further improve the amount of self-consumed energy for the building.

Cillari et al. [32] found that only 40% of the produced electricity from the PV panels were consumed by the building without energy storage. By introducing energy storage this value increased to 97%. A study performed by Liu et al. [14] claimed that by adjusting the angle of the PV panels, the self-consumption in some of the analyzed buildings were increased to 100%. The mentioned thesis further implied that maximizing the selfconsumption might provoke a reduction of produced electricity. It is therefore important to find appropriate weighting factors for the evaluated KPIs, as an optimization of one KPI might reduce another important KPI.

Kayo et al. [33] concluded that zero emission neighborhood (ZEN) is favorable compared to ZEBs in order to increase the self-consumption. The concept of a ZEN differs from a ZEB as it is not as limited in size, shape and availability of on-site energy generation as a single ZEB. Merging the energy generation systems for several buildings may also reduce the total investment cost, compared to investing in energy generation system for each ZEB in the neighborhood. The possibility of increased self-consumption for ZEN is caused by the possibility to transfer surplus electricity between buildings. An office building, a hospital, a hotel and a shopping centre in Japan was studied separately and in clusters. By applying three different control strategies for the combined heat and power (CHP) generator, the amount of annually primary energy use was studied. It was discovered that with any combination of buildings, the annually used energy decreased, compared to the energy used when the buildings were operating separately. Furthermore, a reduction in surplus electricity was found, hence an increasing in self-consumption.

Lopes et al. [30] performed a research on cooperative nZEBs, confirming that by connecting nZEBs the total self-consumption could be increased. They found an improvement of 18-20% in self-consumption in the community, compared to the respective values for each building separately. The authors assumed the increase to be caused by the difference in load profile for each building, as well as an increased amount of loads manageable by the DSM system in the community. Furthermore, the potential of increased amount of available generated energy is mentioned as one of the causes.

2.4.3 Optimizing peak load

A third optimization objective found in literature is the minimization of peak load. A new price regulation for grid costumers in Norway is to be implemented by the end of 2022 [34], encouraging reduction in peak load. Today Norwegian grid costumers pay a constant price for grid rent, in addition to the price for the amount of electricity imported from the grid. The new prices suggests that costumers also pay according to the peak load of the imported electricity. This makes minimization of peak load more valuable.

Comodi et al. [35] investigated a multi apartment ZEB in Italy. They found that the TES provided peak shaving potential, as heat energy could be stored and used during peak hours. The building was equipped with both a battery for the PV production and TES solutions. Though both the battery and the TES increased the energy flexibility potential of the building, only the TES was profitable at the current price conditions. The article expresses the need for energy storage solutions to increase the energy flexibility of a building, especially regarding the peak shaving potential. It further emphasise the importance of considering economy when analysing optimal operation of ZEBs.

2.4.4 Optimizing energy use

The last evaluation criteria to be discussed is the minimization of energy use. During the design phase, research on minimization of energy use is based on choices of technical installations and materials, while research during operational phase focus on how these components can work together in an optimal way. As ZEBs are considered one of the solutions to reducing the world wide energy use, more research on minimization of energy use based on optimal operation is to be expected in the coming years.

A problem that might occur in ZEBs is that the estimated self-sufficiency during design phase does not match the one during operational phase. This can be due to, for instance, mismatch between estimated and actual occupancy behaviour [36]. Mavrigiannaki et al. [37] performed a research to confirm the level of ZEN on a neighborhood in Italy. It showed that there was a difference in the gap between estimated and monitored electricity consumption. While the gap was significant during heating season, it was reduced during cooling season. The authors concluded that this was caused by individual preferences of the occupants controlling the HVAC system. However, the research also found that though there was a monthly mismatch between consumption and production, the yearly electricity consumption did not exceed the electricity production.

Zhou et al. [38] did a study on an existing ZEB in China. In contrast to Mavrigiannaki et al. they discovered that the gap between energy use and on-site generation of energy had increased compared to the estimated energy balance, to a level where the building no longer qualified as a ZEB. The main reasons found for the mismatch was variations in the weather file used for the simulation compared to the measured data, causing less produced electricity from the PV panel than predicted. The second reason was the lack of competence among the occupants to use the GSHP system in an optimal way. A slight difference in building function and equipment from the design phase to the operational phase led to further gap between predicted and measured energy use. This discovery shows the need for not only energy simulations during the design phase, but also monitoring during the operational phase as those values might differ significantly.

Though the main focus on reduction of energy use is during the design phase, the transition from design to operational phase is of importance to reach the ZEB requirements. Ferrara et al. [9] points out the importance of considering the relation between variables in a building to achieve optimal operation. The utilization of reliable energy models must therefore be stressed in attempts to reduce the gap in performance between the design phase and the operational phase.

3 Methodology

During the project work related to this master thesis, an attempt was done to gather measured data on electricity consumption and production from casaZero. Due to difficulties in accessing sufficient amount of data, it was decided to utilize a simulation based method to assess the energy flexibility potential of casaZero. The available data was used to calibrate the simulation model. This chapter includes description of the chosen simulation software. It describes the building utilized in this research as well as the three KPIs evaluated in this research. At last this chapter includes a section on the process of assessing the energy flexibility potential of casaZero.

To decide the simulation software utilized in this study, it was decided to emphasize the following characteristics:

- time efficient
- user friendly
- ability to simulate relevant outputs for energy flexibility analysis
- transparent

Based on these characteristics it was decided to utilize the simulation software IDA ICE.

3.1 IDA ICE

IDA ICE is a Building Performance Simulation (BPS) tool developed by the Swedish company EQUA Simulation AB. A mathematical model of a building can be made in IDA ICE by implementing physical parameters of building envelope and technical equipment in the building, such as ventilation system and heating mechanisms. The model is then used to simulate indoor climate and energy use. IDA ICE support import from Building Information Modeling (BIM) tools via Industry Foundation Classes (IFC) like AutoCAD, ArchiCAD and Autodesk Revit. Physical properties of a building can therefore be directly implemented from these softwares [39]. To find the influence of different operational strategies on the GSHP concerning three KPIs, calculations and experimental methods would be too time demanding. A one year simulation period of casaZero was performed in less than two hours, by utilization of IDA ICE. This enabled the possibility to try different operational strategies for the GSHP, without spending several years measuring data from the building.

The software is intuitive and user friendly. Hence, it is easy to use if the modeling skills in advance are limited. Furthermore, IDA ICE offers a wide range of output parameters such as temperatures, unmet heating hours, CO2 concentration etc. This makes it possible to analyse the indoor climate for each zone in detail. An energy report is given for each simulation, including the electricity consumption and production as well as import and export of electricity to the grid. Figure 6 shows an excerpt of the energy report from a simulation in IDA ICE. The electricity consumption is divided into different types of loads. This is of importance when assessing the energy flexibility potential of a building. Another advantage of utilizing IDA ICE is the transparency of the software, enabling the user to inspect every detail of the model.

		Total		Peak demand	
		kWh	kWh/m ²	kW	Time
	Lighting, facility	1519.2	6.5	0.2538	18 Dec 07:47
	Electric cooling	232.0	1.0	1.779	01 Aug 17:04
	HVAC aux	1316.2	5.6	0.8144	15 Jul 17:57
	Electric heating	1334.1	5.7	7.899	09 Feb 04:43
	Total Facility	4401.5	18.8		
	Equipment, tenant	1380.3	5.9	0.2263	18 Dec 07:48
	Total Tenant	1380.3	5.9		
	PV production	-10392.5	-44.5	-9.272	15 Jul 13:43
	CHP electricity	0.0	0.0	0.0	
	Total Produced	-10392.5	-44.5		
	Electricity, balance	-4610.7	-19.7		



Figure 6: An excerpt of the delivered energy report given when running simulations in IDA ICE.

For evaluation of cost and income according to the delivered energy, IDA ICE provides settings for energy contracts. These contracts enables the user to implement prices for imported and exported energy, depending on the energy carrier. CO2 emission can also be added per volume of each energy carrier. This enables calculations on both the economical aspect and the environmental impact of the chosen solution. As cost was chosen as one of the KPIs analysed in this thesis, the potential of economical calculations was of importance when choosing the simulation software.

With the newest version of IDA ICE, IDA ICE 5, optimization tools are included in

the software. In the newest version both GenOpt and AutoMOO can be initiated from IDA ICE. While GenOpt needs to be available as an external software on the computer, optimization utilizing AutoMOO can be done exclusively in IDA ICE. Figure 7 shows a chart flow of the optimization process utilizing AutoMOO. As described in the figure text, output data are given multiple times during an optimization. When the optimization process is done, the data is displayed in a diagram showing the results for every run during the optimization. The optimization function in IDA ICE enables optimization of several parameters at a time.



Figure 7: The chart flow for a process in IDA ICE, aiming to maximize the selfconsumption. Black lines represent processes that are only done once, while the red lines represent repetitive processes that are done several times during an optimization.

During the work of this thesis, the development of IDA ICE 5 was still ongoing. Therefore, a beta version of the software was used. The optimization function was not capable of performing the planned optimization simulation due to a bug in the utilized version of IDA ICE. As a result of this, a manually implementation of different schedules for operation of the GSHP was implemented to evaluate its influence on the chosen KPIs. The procedure is described in the following sections.

3.2 CasaZero

The ZEB analysed in this thesis is a renovated residential building from the 1960s located in Kolsås, close to Oslo. The building consists of 3 levels, with two floors above and one below the ground level. The total heated area of the building is 233.6 m^2 .

The building was divided into 21 zones, representing the rooms of casaZero. In the actual building, the living room area and the hall/entrance area on the 1. floor is open as shown in Figure 8. To reduce the size of the zones and increase control, the IDA ICE model was made with separate zones for the hall and the living room. The floor plan made in IDA ICE for each level is displayed in Figure 9. In casaZero there is an open solution between the entrance hall and the living room on the second floor. To simulate this in IDA ICE, zones where chosen differing from the actual zones of the building. The hall on the 1. floor was divided into three zones due to difference in room height. One zone representing the part of the hall covering the 2. floor. Another zone representing the hall in the 1. floor, and the last zone representing the entrance area. To simulate the actual circumstances, the hall zone on the 2. floor was equipped with a large opening in the floor, for heat to circulate as if there were no construction between the two zones. For the separated zones, openings were located on the walls between the zones to simulate the actual open area. Furthermore, the stairs to the basement represents an opening between the 1.floor and the basement in the actual building, this however, was not considered in the model. The exterior doors and the remaining interior doors were kept close during the simulation.



Figure 8: The actual floor plan for the 1.floor in casaZero drawn by Snøhetta.



Figure 9: The floor plan for all three levels in casaZero made in IDA ICE.

The orientation of casaZero is shown in Figure 10. No shading objects, such as threes or neighbouring buildings, were considered in the model.



Figure 10: The orientation of casaZero.

The south side of the roof of casaZero is covered with 50 m² of PV panels with a nominal effect of 0.18. The PV panels in the model were placed separate from the roof to simulate the measured amount of electricity produced. Placing PV panels on the roof did not provide enough electricity compared to the actual amount measured in the building. They were however, mounted with a 36 degree of inclination to match the angle of the roofs.

3.2.1 Building envelope

The input parameters for the envelope was calculated based on chosen materials and depth of the construction. Table 1 shows an overview of the physical parameters of the building envelop.

Description	Value	Unit
U-value outer wall	0,11	$[W/m^2K]$
U-value roof	0,12	$[W/m^2K]$
U-value basement floor	0,1	$[W/m^2K]$
U-value for basement wall	0,14	$[W/m^2K]$
U-value outer door/window	0,8	$[W/m^2K]$
Normalized value of thermal bridge	0,08	$[W/m^2K]$
Area outer wall	173	$[m^2]$
Area roof	108	$[m^2]$
Area basement floor	83	$[m^2]$
Area outer door/window	36	$[m^2]$
Total heated area	226	$[m^2]$
Heated air volume	490	$[m^3]$
Leakage number	1	[1/h]

Table 1: Overview of the physical parameters of the building.

3.2.2 Climate and location

As IDA ICE offers a limited amount of climate files and locations, Oslo/Gardemoen was chosen as it was the closest available location to Kolsås. The weather files available in IDA ICE are "typical" weather files, established based on several years of data gathering and calculations of mean values. Due to this, the weather files do not reflect a specific year. It was therefore decided to make an attempt at establishing a weather file based on downloaded weather data from shinyweatherdata.com. The established file was, however, not readable in IDA ICE, so the "typical" weather file was therefore chosen. Figure 11 shows the ambient air temperatures used for the one year simulation. The temperature file shows ambient air temperatures above 25 only in July and August, while the lowest temperatures are found in January. The wind profile was set to suburban.



Figure 11: The ambient air temperatures used for simulation in IDA ICE.

3.2.3 Internal gains and human behaviour

As mentioned in the literature review, the occupancy behaviour might influence the electricity consumption greatly and can therefore cause a source of error in the estimations. As the model was calibrated based on measured values for produced and consumed electricity for 2021, the occupancy behaviour might differ from an average year, due to more frequent utilization of home office and home schooling during the period with covid restrictions. However, the input values for lighting and equipment were based on standard values presented in SN/TS 3031:2016 [40]. The SN/TS 3031:2016 schedules for equipment and lighting are shown in Figure 12. The percentage of supplied heat energy to the zones compared to the electricity consumed by lighting and equipment was set to respectively 100 and 60 according to SN/TS 3031:2016.

Effect [kW/m^2]



Figure 12: The schedules for lighting and equipment in residential buildings based on the schedules found in SN/TS 3031:2016.

For the occupancy it was decided to implement two different schedules, one for the bedrooms and another for the living rooms of the building. The occupancy of the bedrooms was set to 1 during the night and 0 during the day while the living rooms had the opposite occupancy schedule.

3.2.4 Ventilation and AHU

CasaZero is equipped with a balanced ventilation system of the type FLEXIT Nordic S4 with a heat recovery between 80% and 85%. A graph was used in the model to decide the set point temperature for supply air. When the ambient air temperature was lower than 0 degrees, the supply air temperature was set to 20. With ambient air temperatures between 0 and 10 degrees, the supply temperature was 18 and for ambient air temperatures above 10 degrees, the supply temperature was set to 16 degrees. The schedule was established to simulate electricity consumption and temperatures matching the measured values from the building.

3.2.5 The plant

The main source of heat supply to casaZero is through a F1155 NIBE GSHP, providing heat for both the DHW system and the space heating system. The plant in IDA ICE included the GSHP system and the energy and heat distribution system. Figure 13 shows part of the plant for casaZero. The PV panels are not shown in the plant as they were implemented to the electrical system of the model. The average use of DHW was set to 60L/day for each of the 5 occupants of the building. A brine to water heat pump with ground heat exchange was chosen in IDA ICE. The capacity of the GSHP was set to 6kW and the COP was set to 4.



Figure 13: Part of the plant established in IDA ICE for casaZero.

For the space heating system, the basement and 1st floor was equipped with water borne heating coils integrated in the floor. The space heating system was divided into three coils supplying different zones of the building, one for the bathrooms and the laundry, one for the basement and one for the 1.st floor. Each coil was provided with pumps and a unique set point temperature. A cold water tank was implemented in the model to simulate the ability to use the GSHP for cooling, during days with a cooling need.

The water tank was set to 224L. Figure 14 shows the heat and electricity flow between equipment in casaZero. In this chart, electricity for pumps in the space heating system is included in "Other el. specific loads". The battery of the EV is also included in el. specific loads. As shown in the chart, excess electricity from the PV panels are transported back to the grid when the el. specific loads and electricity for GSHP and ventilation is covered. The heat transfer from the heating demand to the ventilation system represents the heat from the heat recovery of the AHU.


Figure 14: The technical equipment of casaZero including the electricity and heat flow between the equipment.

3.2.6 Electricity prices

The schedules for electricity prices were established based on data from NordPool [41]. To estimate the selling price, the spot price for ten randomly chosen days for each month were studied in detail to find an average price curve for each month. The twelve curves were implemented in the electricity contract function in IDA ICE. The prices for 2021 were utilized to establish the curves. The curve for the average spot prices for December 2021 is shown in Figure 15.





To find the cost for bought electricity Equation 1 was used, where C_{tot} is the total

cost in NOK/kWh, C_{const} is the constant grid fee, C_{var} is the variable grid fee, C_{tax} is the consumer taxes and C_{spot} is the spot price for electricity. The constant grid fee was set to 115 NOK/month, while the variable grid fee was 44.8 øre/kWh. The consumer taxes for 2021 was 16.69 øre/kWh. The prices for Oslo region was chosen for both selling and buying prices.

$$C_{tot} = C_{const} + (C_{var} + C_{tax} + C_{spot}) * 1.25$$
(1)

The Norwegian government decided to give compensation for high electricity prices for the period from December 2021 to March 2022 [42]. However, this is not included in the electricity prices implemented in the model.

3.3 Key Performance Indicators

To assess the energy flexibility potential of casaZero, three KPIs were chosen as evaluation criteria. The chosen KPIs in this thesis were electricity import peak, cost associated with electricity consumption and self-consumption. For all three KPIs a set of common criteria were made. The first criteria was made as a result of the ZEB definition, considering the energy balance. As shown in Equation 2, the total electricity consumption could not exceed the total electricity production for the one year period. For each simulation this was checked to ensure a positive energy balance.

$$El^{imp} < El^{exp} \tag{2}$$

 El^{imp} and El^{exp} is respectively total electricity imported and total electricity exported during a period of one year.

The second criteria was the indoor thermal comfort. It was chosen to evaluate the thermal comfort by using the percentage of total occupancy hours with thermal dissatisfaction, referred to as the percentage of people dissatisfied (PPD). This was done by separately summing up the total occupancy hours for all zones and the total hours of thermal dissatisfaction. The two numbers were then implemented in Equation 3.

$$PPD = \frac{h^{PD}}{h^{occ}} \tag{3}$$

Where h^{PD} is the hours of people dissatisfied and h^{occ} is the occupancy hours. Only schedules with PPD<10% were qualified as satisfactory based on the standards given by NS-EN ISO 7730:2005 [43].

3.3.1 Self-consumption

The self-consumption for each model simulated in IDA ICE was calculated using Equation 4.

$$SC = \frac{El^{prod} - El^{exp}}{El^{prod}} \tag{4}$$

Where SC is the self-consumption, El^{prod} is the produced electricity and El^{exp} is the exported electricity for the hole year. The exported electricity was found by subtracting the produced electricity from the total consumed electricity for each time step of the simulation period.

3.3.2 Peak load

The second KPI evaluated in this thesis was the imported electricity peak load. The equation used for calculating the total imported electricity for each time step is shown in Equation 5. A code was established in Python to find the highest value of imported electricity during a year. The code is displayed in Appendix 1.

$$El_t^{imp} = El_t^{heat} + El_t^{eq} + El_t^{hvac} + El_t^{light} - El_t^{prod}$$
(5)

 El_{t}^{imp} is the imported electricity, El_{t}^{heat} is the electricity consumed by electric heating, El_{t}^{eq} is the electricity consumed by equipment, El_{t}^{hvac} is the electricity consumed by the HVAC system, El_{t}^{light} is the electricity consumed by lighting and El_{t}^{prod} is the produced electricity, with t representing the time step of the consumption and production for each hour of the simulation period.

Due to the planned fees for electricity peaks mentioned in the theory chapter, a Python script was established to find the electricity peak for each month of the year. The script is displayed in Appendix 2.

3.3.3 Cost

For the cost analysis the total cost calculated by IDA ICE, based on the implemented costs and earnings from electricity import and export, was used.

3.4 Assessing the energy flexibility potential

An overview of the procedure followed in this thesis for assessing the energy flexibility potential of casaZero is shown in Figure 16. Base case represents the building when the GSHP is regulated only by a thermometer in the DHW tank. After running the first 20 schedules, Case 1, Case 2 and Case 3 were established, containing the three schedules with best results for respectively self-consumption, peak load and cost.



Figure 16: Overview of the methodology used in this thesis.

A total of 65 schedules were implemented and evaluated during the process of finding the best results. Detailed information on each of the four cases is found in chapter 4.

4 Case study

The first section of this chapter presents Base case. The second section presents the implementation of the 20 schedules utilized when deciding Case 1, Case 2 and Case 3. Last is a sub-chapter describing the three cases and the sensitivity analysis performed on each case. Furthermore, it describes the process of finding the schedule with best overall result.

4.1 Base case

Base case was established to simulate the energy use for a one year period, without schedules controlling the GSHP. It was established as described in the methodology chapter. The GSHP was regulated by a thermometer in the DHW tank. Base case also included an electric boiler for top heating to assist the GSHP during the days with highest heating need. The results found for Base case was used as a starting point to compare the results found for each of the schedules implemented to the GSHP.

4.2 GSHP schedules

To control the GSHP a set of 20 schedules were established. They were implemented in the plant of the model, controlling the flow from the GSHP to the DHW tank. A simulation was run for each schedule. The schedules had a 24 hour time span as shown in Figure 17, and did not vary between weekdays and weekends, nor did they change with the seasons. The Top heating shown in Figure 13 was removed, but the thermometer in the DHW tank was kept. Hence, the GSHP was only operating when both the schedules and the thermometer allowed it. All 20 schedules are shown in Appendix 3. The schedules were named S1, S2, ..., S20.



Figure 17: GSHP schedule for S1.

4.3 Case 1, Case 2 and Case 3

The 20 schedules described in 4.2 were evaluated based on the results for each KPI. The schedules with best results were categorized as follows:

- Case 1: the three schedules with highest self-consumption.
- Case 2: the three schedules with lowest yearly peak load.
- Case 3: the three schedules with lowest yearly cost.

A sensitivity analysis was performed on each schedule in the three cases by making minor changes in the schedules. For each case a total of 15 new schedules were established, 5 for each schedule in the case. A total of 45 new schedules were established during the sensitivity analyses of the three cases. The new schedules were named after the schedules in each case, for example, S1(1), S1(2), ..., S1(5).

After finding the optimal schedule for Case 1 an additional load was implemented in IDA ICE to simulate the battery of an EV. This was done to find the effect on selfconsumption provided by the EV for the optimal schedule. A schedule was added to the load to control the charging of the battery. It was set to charge from 10 to 13 every day. The charging period was decided based on when the electricity production was highest. The vehicle was assumed to consume 2kW electricity for each mile, and the yearly driving distance was set to 11200km. This gave a daily driving distance of approximately 3 miles. Hence, a total of 6kWh per day.

In addition to finding the schedule providing the best result for each KPI, the schedule with the highest overall score was found, considering all 65 schedules. To find the schedule with best overall score when weighting each of the KPIs equally, points between 0 and 1 were given for each KPI for all schedules with acceptable thermal comfort. This was done by finding the most optimal result for each KPI and divide the respective value for each schedule by the optimal value. The score for each KPI were added and the schedule with a value closest to 3 was designated as the best overall model.

5 Results and Analysis

This chapter presents and analyses the results found in this thesis. First the results from Base case are presented, followed by a sub chapter on identifying Case 1, Case 2 and Case 3. The results from the sensitivity analyses of each of the three cases are presented in separate sub chapters. The last part of this chapter describes the schedule with the best overall result.

5.1 Base case

The energy and indoor climate simulation for the one year period with Base case gave the following main results:

- Total produced electricity: 10 479 kWh.
- Total consumed electricity: 9559 kWh.
- Total purchased electricity: 5732 kWh.
- Total exported electricity: 6653 kWh.

Figure 18 and Figure 19 shows the load profile for the three days in a row with highest total electricity demand for each of the four seasons. While the load profile for winter season has two significant peaks for each day, these peaks are less prominent for the other seasons. It can also be seen that while the electric heating is the most varying factor for autumn, winter and spring season, the electricity consumption from electric heating is rather stable for the summer load profile. The heating, ventilation and air conditioning (HVAC), however, varies more in the load profile for August. This causes higher peaks in the afternoon from 15 to 18 when the cooling need is high. As the total electricity load is significantly reduced during the summer season, the energy flexibility potential measured in self-consumption, peak load reduction and cost reduction is at its lowest during the summer, unless, for instance, another shiftable load is introduced to the building.



Figure 18: Load profile for the 72 hour periods with highest electricity consumption for winter and spring season for Base case.

As the equipment and the lighting follows set schedules for the entire year, the electricity consumption for these two loads does not vary between days or seasons. This is confirmed in the load profiles found for each season.



Figure 19: Load profile for the 72 hour periods with highest electricity consumption for summer and autumn season for Base case.

The self-consumption found for Base case was 0.379. Hence, more than half of the

produced electricity was exported to the grid during the simulation period. The yearly peak load was found to be 7788W. While the monthly peak loads from November to March varied between approximately 5000W and 8000W, the monthly peak loads from April to October were stable below 2000W. The yearly cost for Base case was 9924.3NOK.

5.2 Identifying Case 1, Case 2 and Case 3

The results from the simulations with the first 20 schedules are shown in Figure 20. Case 1 is identified as S6, S13 and S16, marked in green. Case 2 is marked in yellow as S7, S9 and S20. Case 3 is marked in blue as S2, S12 and S13. It is noticeable that both S13 and S16 have a PPD above the acceptable limit. They were, however, not rejected as the sensitivity analysis would effect the PPD value.

Schedule	SC [-]	Peak in [W]	Cost [NOK]	PPD [%]
Base case	0,37913978	7788	9924,3	9,81
S1	0,342074	5215	9419,1	9,75
S2	0,25855508	5831	9213,9	9,82
S3	0,32352651	5671	9692,3	9,72
S4	0,33175741	5328	9610,5	9,7
S5	0,25859354	5877	9791,5	9,82
S6	0,39836312	5897	9470,6	9,79
S7	0,32640015	4700	9712,3	9,72
S8	0,32071676	5204	9666,2	9,72
S9	0,32886731	4898	9643,3	9,74
S10	0,33634469	5189	9506,4	9,78
S11	0,25856105	5970	9279,1	9,74
S12	0,25855362	5610	8962,2	9,84
S13	0,43269587	5445	8751,7	10,5
S14	0,39073711	5930	9296,7	9,81
S15	0,35545245	5570	9358,6	9,67
S16	0,43794003	5780	9278,1	10,2
S17	0,34409583	5498	9237,2	9,77
S18	0,32001374	5009	9545,3	9,78
S19	0,35301498	4959	9515,8	9,77
S20	0,34216941	4949	9490,5	9,78

Figure 20: Result from the first 20 schedules implemented in the simulation model. The three best results considering each KPI is marked in green, yellow and blue.

The schedule with highest score for cost minimization, peak load reduction and selfconsumption from the first round of simulation were respectively S13, S7 and S16. These are displayed in Figure 21. In addition to having the lowest cost compared to the 20 first models, S13 also had one of the best results for self-consumption. This schedule was therefore further improved according to both self-consumption and cost.



Figure 21: The three schedules with highest score of the 20 first schedules. S13 gave the lowest cost, S16 gave the highest self-consumption, while S7 gave the lowest imported electricity peak.

5.3 Case 1

The 15 schedules established based on Case 1 is presented in Figure 22. The row written in red shows the schedule with the highest self-consumption and acceptable thermal comfort. The schedules from Case 1 are marked in green. All 15 schedules are displayed in Appendix 4.

Schedule	SC [-]	Peak load [W]	Cost [NOK]	PPD [%]
S6(1)	0,38881859	5868	9538,1	9,77
S6(2)	0,38829516	5811	9579,9	9,76
S6(3)	0,41753634	5995	9476,5	9,87
S6(4)	0,41346548	5981	9518,9	9,88
S6(5)	0,37671711	5616	9482,1	9,73
S13(1)	0,43698146	5556	8870,9	10,37
S13(2)	0,43854168	5509	8989,6	10,43
S13(3)	0,43466116	5549	9107,2	10,34
S13(4)	0,42937036	5591	9221,5	10,28
S13(5)	0,42786048	5305	8795,5	10,5
S16(1)	0,43295207	5758	9378	10,15
S16(2)	0,42786149	5755	9466,1	10,11
S16(3)	0,43092801	5873	9405,4	9,98
S16(4)	0,42692416	5845	9496,1	10
S16(5)	0,42230581	5853	9554,2	9,99
S6	0,39836312	5897	9470,6	9,79
S13	0,43269587	5445	8751,7	10,5
S16	0,43794003	5780	9278,1	10,2

Figure 22: An overview of the main results from sensitivity analysis of Case 1.

As seen in the right column of the figure, the PPD is not satisfactory for many of the schedules. This is caused by the need to increase the electricity consumption during PV production hours. By focusing most of the heating hours to the middle of the day, the risk of not meeting the heating need during evening and morning increases. As shown in Figure 21, for both S13 and S16 the GSHP is on only during the middle of the day. For S13 this resulted in all five new schedules not reaching acceptable value of PPD. For S16 however, some of the new schedules have acceptable indoor thermal climate. Of the schedules with PPD<10%, S16(3) has the highest self-consumption with an increasing of 13.7%, compared to Base case. Furthermore, S16(3) has a 25.1% decrease in imported electricity and a 12.9% decrease in exported electricity. This can be considered a significant reduction of interaction with the grid. For this schedule, both cost and peak load decreased, compared to Base case.

Figure 23 shows the GSHP schedule for S16(3). Though the working hours of the GSHP is only extended with one hour compared to the S16 schedule, the effect on the PPD is valuable as it reaches an acceptable level. However, the increased number of operational hours of the GSHP resulted in a slightly decrease in self-consumption, compared to S16. This is due to operation distributed on one more hour outside the PV production hours. It was found that the annual electricity consumption for S16(3) was reduced by 6.1%, compared to Base case. Hence, with smart controlling of the GSHP the annual electricity consumption.



Figure 23: GSHP schedule for S16(3).

To assess the difference in self-consumption between different days of the year, the self-consumption for some selected days were studied. The electricity production and consumption for the first day of January, April, August and November are displayed in Figure 24. The self-consumed electricity is the area below both the production curve and the consumption curve. It is obvious from the figure that 1. August has the lowest self-consumption compared to the three other days. It shows a gap of almost 7kW at 12 o clock, when the heating and cooling demand is low, causing a low electricity consumption. The curve for 1. January and 1. November shows a significantly higher self-consumption. As seen in the figure, the self-consumption could be further increased by postponing the morning peak load. An attempt at this was done in S16(1) where the GSHP could operate from 9 to 19. Though this improved the self-consumption, the PPD increased to above 10.



Figure 24: Electricity production vs electricity consumption for four selected days during the one year simulation period with S16(3).

A comparison of the seasonal differences was investigated comparing Base case and

S16(3). The results are shown in Table 2. Though the self-consumption for all seasons were improved for S16(3), the largest increase was found during the winter when the heating need is at its highest. The lowest increase was found during the summer. For both Base case and S16(3), the self-consumption value for summer is closest to the overall value for the entire year. This implies that to significantly increase the self-consumption an increase in self-consumption during summer is necessary. This can be done by introducing a controllable load to match the high electricity production during summer, or by introducing an electric battery for electricity storage.

Table 2: The self-consumption for S16(3) and Base case for each season and for a hole year.

	Base case	S16(3)
Winter	0.459	0.603
Spring	0.348	0.391
Summer	0.372	0.404
Autumn	0.413	0.478
Yearly	0.379	0.431

By introducing an additional load representing the EV, the yearly self-consumption increased to 0.581. This is a 53.3% increase from Base case and a 34.8% increase from S16(3). Hence, the ability to utilize smart charging of an EV increases the energy flexibility potential of the building. However, it should be mentioned that the peak hours of electricity production coincides with normal office hours. The EV might therefore not be connected to the charging system of the house at this time of the day. The import peak load and the yearly cost were increased to respectively 5904W and 11 269NOK.

5.4 Case 2

The 15 schedules established based on Case 2 is presented in Figure 25. The row written in red shows the schedules with the lowest yearly peak load and acceptable thermal comfort. The schedules from Case 2 are marked in green. All 15 schedules are displayed in Appendix 5.

Schedule	SC [-]	Peak load [W]	Cost [NOK]	PPD [%]
S7(1)	0,37692517	4145	9790,5	9,81
S7(2)	0,36270583	4411	9734,4	9,77
S7(3)	0,353194	4355	9769	9,75
S7(4)	0,38211529	4627	9660,7	9,8
S7(5)	0,32603356	4842	9766,2	9,78
S9(1)	0,34732003	4798	9654,8	9,76
S9(2)	0,30728642	3896	9763,8	9,76
S9(3)	0,35052978	4885	9673,5	9,74
S9(4)	0,32448456	4901	9750,9	9,71
S9(5)	0,34368623	4510	9793,2	9,76
S20(1)	0,37670432	4238	9808,4	9,79
S20(2)	0,34504328	5021	9470,8	9,71
S20(3)	0,37449012	5380	9405	10,09
S20(4)	0,40187575	5447	9349,9	10,19
S20(5)	0,34074366	4989	9398,9	9,77
S7	0,32640015	4700	9712,3	9,72
S9	0,32886731	4898	9643,3	9,74
S20	0,34216941	4949	9490,5	9,78

Figure 25: An overview of the main results from the sensitivity analysis of Case 2.

Based on the results in Figure 25, schedule S9(2) gives the lowest result for yearly peak load. The schedule for S9(2) is displayed in Figure 26, showing a constant value of 0.5 except from 8 to 11 and 15 to 18 when the GSHP is off.



Figure 26: GSHP schedule for S9(2).

Further analyses of the results were done utilizing the Python code for monthly peak load. Figure 27 gives an overview of the peak load for each month in Base case compared to S7, S9, S7(1) and S9(2). The months from April to October have a low monthly peak load. This is due to the reduction in heating demand as well as an increase in electricity production. The peak for each of the mentioned months was found between 6 and 9 in the evening for all four schedules. From Figure 29 it can be seen that the electricity production is low at 18 and zero at 21. If the import electricity peaks were shifted to the period when the electricity production was larger, the monthly peak loads might decrease further, and the yearly self-consumption might increase. The risk of increasing the value of PPD would, however, be significant as hours with large electricity production are the hours with less heating need.

Though the S9(2) schedule has the lowest yearly peak load, the figure shows that S7(1) puts a lower pressure on the grid for all months except January and March. In April the peak load for S7(1) was reduced by 37.4% compared to S9(2), and the average peak load reduction for the 12 months was 11.3%. It can also be noted from the figure that Base

case has a low peak load for the months from April to October. However, the peak loads during the winter months are about doubled compared to the ones for S7(1) and S9(2). Considering the new electricity fees it can be assumed that S7(1) is the preferred schedule for a house owner. S9(2) is the preferred schedule in regard to dimensioning of the grid, as it has the lowest maximum peak load.



Figure 27: Monthly peak load for Base case, S7, S9, S7(1) and S9(2).

The three days for each season with highest electricity consumption are displayed in Figure 28 and Figure 29. It can be noted that the dimensioning load for summer and autumn falls on the same date as for Base case. The figure shows a slightly lower self-consumption value for S9(2), due to the off-periods for the GSHP. Furthermore, it can be seen from Figure 28 and Figure 29 that the dimensioning load for winter and spring season is significantly lower for S9(2), compared to Base case. This is reasonable as the schedule for S9(2) limits the highest peaks. However, the off periods of the GSHP is compensated with higher peaks during operating hours of the GSHP. This causes slightly higher peaks during the summer season for S9(2), compared to Base case.

The compensation for the off-periods of the GSHP might also be one of the reasons for S7(1) having the lowest monthly peak load for 10 out of 12 months. As the schedule for S7(1) is a constant value of 0.5, the load from the GSHP can be distributed on 24 hours. A way of finding a better schedule for lowering the average monthly peak load could therefore be to find the lowest constant value for regulating the GSHP, without compromising the thermal comfort of the building.



Figure 28: Load profile for the 72 hour periods with highest electricity consumption for winter and spring season for S9(2).



Figure 29: Load profile for the 72 hour periods with highest electricity consumption for summer and autumn season for S9(2).

Though S7(1) is considered the best solution regarding monthly peak load, S9(2) has the highest reduction in yearly peak load with a reduction of 50%. The equivalent number for S7(1) is 47%. Hence, the S9(2) is the best model found in this thesis for yearly peak load reduction.

5.5 Case 3

The 15 schedules established based on Case 3 is presented in Figure 30. The row written in red shows the schedule with the lowest yearly cost and acceptable thermal comfort. The schedules from Case 3 are marked in green. All 15 schedules are displayed in Appendix 6.

Schedule	SC [-]	Peak load [W]	Cost [NOK]	PPD [%]
S12(1)	0,26370669	5639	9096,5	9,8
S12(2)	0,2719665	5607	9212,8	9,79
S12(3)	0,28355494	5672	9320,6	9,78
S12(4)	0,26371318	5645	9095	9,8
S12(5)	0,27195027	5665	9210,2	9,79
S2(1)	0,25850281	5905	9366	9,82
S2(2)	0,25718668	6002	9247,1	9,85
S2(3)	0,27722005	5818	9084,2	9,88
S2(4)	0,25850171	5727	9390,5	9,83
S2(5)	0,2585603	5817	9213,1	9,82
S13(1)	0,43211203	5639	9266	10,03
S13(2)	0,43623544	5541	8903	10,41
S13(3)	0,43232551	5634	9266	10,03
S13(4)	0,42677652	5699	9191,3	10,06
S13(5)	0,43474073	5403	9402,8	10,15
S2	0,25855508	5831	9213,9	9,82
S13	0,43269587	5445	8751,7	10,5
S12	0,25855362	5610	8962,2	9,84

Figure 30: An overview of the main results from the sensitivity analysis of Case 3.

It is noticeable that the self-consumption is significantly reduced for all schedules with acceptable PPD. A high self-consumption requires large loads during the hours with PV production, while low costs are achieved when the load is shifted from hours with high electricity prices to hours with low electricity prices. This is demonstrated in Figure 31 using the three schedules displayed in Figure 32. The curves for imported electricity and the total cost found for each of the schedules implies a correlation between load shifting from day time to night time and reduction in cost.

As hours with high electricity prices mostly coincides with hours of PV production, it might be difficult to achieve high self-consumption and low cost at the same time. In an attempt to avoid this issue, the S13 schedule was extended in hours of operation of the GSHP and reduced in maximum operation capacity for the hours with highest cost. The results of this is shown in Figure 30 as S13(1), S13(2),...,S13(5). However, none of the modified schedules for S13 resulted in acceptable PPDs.



Figure 31: Imported electricity for Base case, S12, S2(3) and S20(1).



Figure 32: GSHP schedules for S12, S2(3) and S20(1).

As previously mentioned, the cost estimations and analysis in this thesis does not take into account the new peak fee soon to be implemented in the Norwegian electricity prices. However, the models with acceptable PPD were analysed using the Python script for monthly peak load. A selection of the results are shown in Figure 33. The figure shows that even with similar yearly electricity peaks S12(5) and S2(5) have significant variation in the monthly electricity peaks. This implies that further cost analysis should be done to find the optimal GSHP schedule when considering the new fees. With the cost estimations done in this research, the S12 model was found to have the lowest cost for operation, with a decrease of 9.7% compared to the base model.



Figure 33: Monthly peak load for some selected schedules during sensitivity analysis of Case 3.

5.6 Optimal overall model

As previously mentioned, a correlation between some objectives was found causing, for instance, an increase in one value when decreasing another. The strongest relation was between low cost and low self-consumption. The schedule with highest total score was S7(1). The schedule for S7(1) is shown in Figure 34. Though the schedule did not have the best result found in this thesis for any of the three KPIs, it had above average scores for all three KPIs. Only some of the schedules with unsatisfactory results for PPD had a higher total score.



Figure 34: GSHP schedule for S7(1).

Figure 35 shows that the load profile for the three days with highest electricity consumption during summer season is equal to the load profile found for Base case in Figure 19. This implies that even with S7(1) limiting the capacity of the GSHP, the electricity load is small enough to be covered by the GSHP for each hour of the three days studied in the figure. For the winter season the load profile for S7(1) differs from the load profile for Base case. As the GSHP is not capable of covering the highest peaks, the peaks for S7(1) are wider and lower. Furthermore, the location of the peaks in Figure 35 differ from Figure 18. While the peaks for Base case are only found in the evening or during the night, the electricity consumption for S7(1) also peaks during the day. To increase the self-consumption it is favorable to increase the load during hours of PV production, but with the limitation in peak value the schedule for S7(1) reduces the self-consumption by 0.6%. For the cost it is favorable to shift the load to night hours as the cost is lower during the night. For S7(1) the total yearly electricity consumption is slightly reduced, causing a 1.4% reduction in total cost.

The weighting of each KPI in a multi objective optimization of the operation of a ZEB might vary. However, for most private house owners economics and laws, in addition to thermal comfort, are the major influences on decisions made when designing and operating a building. In this case it was found that the yearly cost for operation of casaZero differed with a maximum of 1000 NOK. Compared to a normal budget for a Norwegian family this can be considered a minor budget burden. However, with the new fees for electricity peaks it is reasonable to assume that for an average residential building owner the reduction of cost and electricity peaks will weight higher than the increasing of self-consumption.



Figure 35: Load profile for the 72 hour periods with highest electricity consumption for winter and summer season for S7(1).

6 Discussion

This section describes some of the sources of inaccuracy and errors found in this thesis. The last part of this section includes an assessment of the DHW tapping schedule utilized in this thesis.

During the work of this thesis, some assumptions and simplifications were made to establish a model in IDA ICE. Examples of such simplifications and assumptions are the implementation of schedules for lighting and equipment. It can be assumed that for each individual house hold the exact consumption will differ from standards given by Norsk Standard. However, this is considered as minor sources of error, as this study aims to find the influence of different operational strategies on the GSHP. Hence the most important load for the results is the electricity consumption for heating. The weather profile used in the model is an estimation based on gathered weather data from previous years, not the exact weather data from 2021. Due to these simplifications, the results found in this thesis might deviate from the actual energy flexibility of the building.

Another source of error found in this thesis was the simplifications done while establishing cost schedules. The work of implementing one cost schedule for each day to simulate the actual daily variations in cost was too time demanding. Due to this, only some days were studied in an attempt to find a suitable cost schedule for each month. The highest peaks and the lowest valleys of the cost curve were therefore not reflected in the simulations.

6.0.1 Schedule for DHW tapping

A source of error was found in the lack of schedule for tapping of DHW. According to NS/TS 3031:2016, the tapping of DHW in a small house is zero from 23 to 5 in the morning and peaks from 7 to 8 and 15 to 19 with a peak value of 1.442 kWh. In the model simulated in this task a constant distribution of tapping of DHW was implemented. A test simulation was therefore run to find the difference in Base case with and without the schedule for tapping of DHW found in NS/TS 3031:2016. The results are shown in Table 3, where BCs is the base model with the correct tapping schedule for DHW.

Model	SC [-]	Peak load	Cost	PPD [%]
		[W]	[NOK]	
Base case	0.37913978	7788	9924.3	9.81
BCs	0.36684745	7823	10052	9.81

Table 3: Main results from Base case and BCs.

As the peaks of the curve for tapping of DHW coincides with the peaks of the cost curve for electricity import, the cost of BCs is naturally increased from the original model. It is reasonable to assume that the cost of each of the 65 models will increase by the same amount by changing only the schedule for DHW tapping. Hence, the cost comparison is not affected by the lack of schedule. The peak loads might increase if the peaks coincides with the peak for DHW tapping. To find the probability of the tapping schedule changing the peak load, the time of the peaks where checked for each GSHP schedule. It was found that the peak load for most of the schedules simulated in the sensitivity analysis of Case 2 peaked during off peak hours compared to the tapping schedule. Hence, the lack of tapping schedule is only a minor influence on the peak load analysis.

The self consumption decreases in the case of BCs compared to Base case. This can be explained studying Figure 36. The figure shows the solar radiation of an 8 day period in January. As shown in the figure, the solar radiation is 0 during the first DHW tapping peak. Hence, the electricity from PV production is 0 from 6 to 8 during the winter season. Parts of the 15 to 19 peak is also during hours of 0 electricity production. In Base case the DHW tapping is equally divided throughout the day, causing a higher consumption during hours of PV production. This gives a slightly decrease in the self-consumption for the BCs model. Though the possibility of inaccuracy in the results increases when not utilizing a proper schedule for tapping of DHW, it is considered insignificant in this thesis. Furthermore, it should be mentioned that the restrictions of the covid year of 2021 might have influenced the tapping schedule to deviate from the standard schedule given by SN/TS 3031:2016.



Figure 36: The solar radiation of a 8 day period in January.

7 Conclusion

65 different GSHP schedules were simulated in this thesis in an attempt to assess how operational strategies of the GSHP affected the energy performance of a residential ZEB. The three KPIs analysed in this thesis were self-consumption, peak load and cost. Table 4 shows the results from the best schedules for each KPI and the schedule with best overall result for all three KPIs. The greatest energy flexibility potential was found in the peak load reduction. The largest reduction in peak load was found to be 50%, but the model providing the lowest yearly peak load was not the model with the lowest average of monthly peak load. It was found in the results that every schedule gave a reduction in peak load compared to Base case. The cost was the least flexible KPI with a maximum reduction of 9.7%. The self-consumption was increased by 13.7% and it was found that the summer season was the most influential season considering reduction of self-consumption.

Table 4: KPI values for the best schedule in each case and the overall best schedule compared to the results from Base case.

Schedule	Base case	S16(3)	$\mathbf{S7}(1)$	S12	$\mathbf{S9}(2)$
SC	0.379 [-]	+13.7%	-0.6%	-31.8%	-19%
Peak load	7788 [W]	-24.6%	-46.8%	-28%	-50%
Cost	9924.3 [NOK]	-5.2%	-1.4%	-9.7%	-1.6%

It was found that without an additional controllable load implemented during the summer months or an electrical battery to store excess electricity, the self-consumption was limited. This was due to a low total electricity consumption during these months. By introducing a load representing the battery of an EV and implementing a schedule for daily charging from 10-13, the self-consumption was increased by 53.3% compared to Base case. However, due to normal office hours this charging schedule was considered unlikely.

When weighting the KPIs equally, the overall best score was found in a schedule with a constant value of 0.5. A relation between the KPIs analysed in this study was found. The schedules with lowest cost also had the lowest values for self-consumption. This thesis concludes that controlling the GSHP in casaZero has a great impact on the energy performance of the building.

8 Further work

Based on the research done in this thesis, a set of suggestions for further research on this topic was made to encourage wider knowledge on optimal and flexible operation of ZEBs.

- Further research on monthly peak load and the cost effect of the new fees for electricity peak is recommended, as the results of this thesis implies a significant variety in monthly peak load.
- Further research on seasonally adapted operation should be done, as this thesis implies a significant change in load profile for the different seasons.
- In this study, a sensitivity analysis was performed on a selection of defined schedules. To investigate the possibility of finding an even better schedule for GSHP operation, more research utilizing optimization tools should be conducted.

References

- IEA. "Buildings A source of enormous untapped efficiency potential." (), [Online]. Available: https://www.iea.org/topics/buildings (visited on 05/04/2022).
- [2] EU, "Directive (EU) 2018/844 of the European Parlament and of the Council of 30 May 2018," Official Journal of the European Union, no. June 2028, 2018, ISSN: 16113349. DOI: 10.1007/3-540-47891-4{_}10.
- [3] —, "DIRECTIVE 2010/31/EU OF THE EUROPEAN PARLIAMENT AND OF THE COUNCIL of 19 May 2010," pp. 13–35, 2010. [Online]. Available: https://eur-lex.europa.eu/legal-content/EN/ALL/?uri=celex% 3A32010L0031.
- [4] M. G. Fikru, J. Atherton, and C. I. Canfield, "Cost-reflective dynamic electricity pricing for prosumers," vol. 35, 2022.
- Y. Parag and B. K. Sovacool, "Electricity market design for the prosumer era," *Nature Energy*, vol. 1, no. 4, 2016, ISSN: 2058-7546. DOI: 10.1038/nenergy. 2016.32.
- [6] K. B. Lindberg, D. Fischer, G. Doorman, M. Korpås, and I. Sartori, "Costoptimal energy system design in Zero Energy Buildings with resulting grid impact: A case study of a German multi-family house," *Energy and Buildings*, vol. 127, pp. 830–845, 2016, ISSN: 03787788. DOI: 10.1016/j.enbuild.2016. 05.063.
- K. B. Lindberg, Impact of Zero Energy Buildings on the Power System -A study of load profiles, flexibility and system investments. 2017, vol. 35, ISBN: 9788232621446. [Online]. Available: http://hdl.handle.net/11250/ 2450566.
- [8] K. Li, L. Pan, W. Xue, H. Jiang, and H. Mao, "Multi-Objective Optimization for Energy Performance Improvement of Residential Buildings: A Comparative Study," *Energies*, vol. 10, no. 2, 2017, ISSN: 19961073. DOI: 10.3390/ en10020245.
- [9] M. Ferrara, C. Lisciandrello, A. Messina, M. Berta, Y. Zhang, and E. Fabrizio, "Optimizing the transition between design and operation of ZEBs: Lessons learnt from the Solar Decathlon China 2018 SCUTxPoliTo prototype," *Energy* and Buildings, vol. 213, 2020, ISSN: 03787788. DOI: 10.1016/j.enbuild. 2020.109824.
- C. Ionescu, T. Baracu, G. E. Vlad, H. Necula, and A. Badea, "The historical evolution of the energy efficient buildings," *Renewable and Sustainable Energy Reviews*, vol. 49, pp. 243–253, 2015, ISSN: 18790690. DOI: 10.1016/j.rser. 2015.04.062.

- M. Hamdy, A. Hasan, and K. Siren, "A multi-stage optimization method for cost-optimal and nearly-zero-energy building solutions in line with the EPBD-recast 2010," *Energy and Buildings*, vol. 56, pp. 189–203, 2013, ISSN: 03787788. DOI: 10.1016/j.enbuild.2012.08.023. [Online]. Available: http://dx.doi.org/10.1016/j.enbuild.2012.08.023.
- [12] G. Lobaccaro, A. H. Wiberg, G. Ceci, M. Manni, N. Lolli, and U. Berardi, "Parametric design to minimize the embodied GHG emissions in a ZEB," *Energy and Buildings*, vol. 167, pp. 106–123, 2018, ISSN: 03787788. DOI: 10. 1016/j.enbuild.2018.02.025.
- J. Yu, C. Yang, and L. Tian, "Low-energy envelope design of residential building in hot summer and cold winter zone in China," *Energy and Buildings*, vol. 40, no. 8, pp. 1536–1546, 2008, ISSN: 03787788. DOI: 10.1016/j.enbuild. 2008.02.020.
- [14] C. Liu, W. Xu, A. Li, D. Sun, and H. Huo, "Analysis and optimization of load matching in photovoltaic systems for zero energy buildings in different climate zones of China," *Journal of Cleaner Production*, vol. 238, p. 117914, 2019, ISSN: 09596526. DOI: 10.1016/j.jclepro.2019.117914. [Online]. Available: https://doi.org/10.1016/j.jclepro.2019.117914.
- [15] R. Wang, W. Feng, L. Wang, and S. Lu, "A comprehensive evaluation of zero energy buildings in cold regions: Actual performance and key technologies of cases from China, the US, and the European Union," *Energy*, vol. 215, 2021, ISSN: 03605442. DOI: 10.1016/j.energy.2020.118992.
- [16] S. N. Al-Saadi and A. K. Shaaban, "Zero energy building (ZEB) in a cooling dominated climate of Oman: Design and energy performance analysis," *Renewable and Sustainable Energy Reviews*, vol. 112, no. June, pp. 299–316, 2019, ISSN: 18790690. DOI: 10.1016/j.rser.2019.05.049.
- [17] T. R. C. on Zero Emission Buildings. "Zeb definitions." (), [Online]. Available: http://www.zeb.no/index.php/en/about-zeb/zeb-definitions (visited on 04/27/2022).
- [18] IEA, Examples of Energy Flexibility in Buildings, September. 2019, ISBN: 9788793250123.
- [19] —, Literature review on energy flexibility definitions and indicators for building clusters, April. 2018.
- [20] M. Farrokhifar, H. Bahmani, B. Faridpak, A. Safari, D. Pozo, and M. Aiello, "Model predictive control for demand side management in buildings : A survey," *Sustainable Cities and Society*, vol. 75, no. September, p. 103 381, 2021, ISSN: 2210-6707. DOI: 10.1016/j.scs.2021.103381. [Online]. Available: https://doi.org/10.1016/j.scs.2021.103381.

- [21] Goran Strbac, "Demand side management : Benefits and challenges \$," vol. 36, pp. 4419–4426, 2008. DOI: 10.1016/j.enpol.2008.09.030.
- [22] R. Luthander, J. Widén, D. Nilsson, and J. Palm, "Photovoltaic self-consumption in buildings: A review," *Applied Energy*, vol. 142, pp. 80–94, 2015, ISSN: 03062619. DOI: 10.1016/j.apenergy.2014.12.028. [Online]. Available: http://dx.doi.org/10.1016/j.apenergy.2014.12.028.
- M. Ferrara, A. Rolfo, F. Prunotto, and E. Fabrizio, "EDeSSOpt Energy Demand and Supply Simultaneous Optimization for cost-optimized design: Application to a multi-family building," *Applied Energy*, vol. 236, no. November 2018, pp. 1231–1248, 2019, ISSN: 03062619. DOI: 10.1016/j.apenergy. 2018.12.043.
- [24] Z. Zheng, Lixiongwang, and N. Hienwong, "Intelligent Control System Integration and Optimization for Zero Energy Buildings to Mitigate Urban Heat Island," *Procedia Engineering*, vol. 169, pp. 100–107, 2016, ISSN: 18777058. DOI: 10.1016/j.proeng.2016.10.012. [Online]. Available: http://dx.doi.org/10.1016/j.proeng.2016.10.012.
- [25] U. I. Dar, I. Sartori, L. Georges, and V. Novakovic, "Advanced control of heat pumps for improved flexibility of Net-ZEB towards the grid," *Energy and Buildings*, vol. 69, pp. 74–84, 2014, ISSN: 03787788. DOI: 10.1016/j.enbuild. 2013.10.019. [Online]. Available: http://dx.doi.org/10.1016/j.enbuild. 2013.10.019.
- M. Ferrara, J. Virgone, E. Fabrizio, F. Kuznik, and M. Filippi, "Modelling Zero Energy Buildings: Parametric study for the technical optimization," *Energy Procedia*, vol. 62, pp. 200–209, 2014, ISSN: 18766102. DOI: 10.1016/j.egypro. 2014.12.381. [Online]. Available: http://dx.doi.org/10.1016/j.egypro. 2014.12.381.
- [27] Y. Zhang, P. Zeng, S. Li, C. Zang, and H. Li, "A novel multiobjective optimization algorithm for home energy management system in smart grid," *Mathematical Problems in Engineering*, vol. 2015, 2015, ISSN: 15635147. DOI: 10.1155/2015/807527.
- [28] K. E. Thorvaldsen, M. Korpås, K. B. Lindberg, and H. Farahmand, "A stochastic operational planning model for a zero emission building with emission compensation," *Applied Energy*, vol. 302, no. February, 2021, ISSN: 03062619. DOI: 10.1016/j.apenergy.2021.117415.
- [29] D. Gudmunds, E. Nyholm, M. Taljegard, and M. Odenberger, "Self-consumption and self-sufficiency for household solar producers when introducing an electric vehicle," *Renewable Energy*, vol. 148, pp. 1200–1215, 2020, ISSN: 18790682.

DOI: 10.1016/j.renene.2019.10.030. [Online]. Available: https://doi. org/10.1016/j.renene.2019.10.030.

- [30] R. A. Lopes, J. Martins, D. Aelenei, and C. P. Lima, "A cooperative net zero energy community to improve load matching," *Renewable Energy*, vol. 93, pp. 1–13, 2016, ISSN: 18790682. DOI: 10.1016/j.renene.2016.02.044.
- [31] A. Milo, H. Gaztañaga, I. Etxeberria-Otadui, S. Bacha, and P. Rodríguez, "Optimal economic exploitation of hydrogen based grid-friendly zero energy buildings," *Renewable Energy*, vol. 36, no. 1, pp. 197–205, 2011, ISSN: 09601481. DOI: 10.1016/j.renene.2010.06.021.
- [32] G. Cillari, A. Franco, and F. Fantozzi, "Sizing strategies of photovoltaic systems in nZEB schemes to maximize the self-consumption share," *Energy Reports*, vol. 7, pp. 6769–6785, 2021, ISSN: 23524847. DOI: 10.1016/j.egyr.2021.09.117. [Online]. Available: https://doi.org/10.1016/j.egyr.2021.09.117.
- [33] G. Kayo, A. Hasan, and K. Siren, "Energy sharing and matching in different combinations of buildings, CHP capacities and operation strategy," *Energy* and Buildings, vol. 82, pp. 685–695, 2014, ISSN: 03787788. DOI: 10.1016/j. enbuild.2014.07.077.
- [34] Lovdata. "Forskrift om endring i forskrift om økonomisk og teknisk rapportering, inntektsramme for nettvirksomheten og tariffer." (), [Online]. Available: https://lovdata.no/dokument/LTI/forskrift/2021-06-10-1904 (visited on 04/27/2022).
- [35] G. Comodi, A. Giantomassi, M. Severini, et al., "Multi-apartment residential microgrid with electrical and thermal storage devices: Experimental analysis and simulation of energy management strategies," Applied Energy, vol. 137, pp. 854–866, 2015, ISSN: 03062619. DOI: 10.1016/j.apenergy.2014.07.068.
 [Online]. Available: http://dx.doi.org/10.1016/j.apenergy.2014.07.068.
- [36] C. Carpino, E. Loukou, P. Heiselberg, and N. Arcuri, "Energy performance gap of a nearly Zero Energy Building (nZEB) in Denmark: the influence of occupancy modelling," *Building Research and Information*, vol. 48, no. 8, pp. 899–921, 2020, ISSN: 14664321. DOI: 10.1080/09613218.2019.1707639. [Online]. Available: https://doi.org/10.1080/09613218.2019.1707639.
- [37] A. Mavrigiannaki, K. Gobakis, D. Kolokotsa, et al., "Zero energy concept at neighborhood level: A case study analysis," Solar Energy Advances, vol. 1, no. July, p. 100002, 2021, ISSN: 26671131. DOI: 10.1016/j.seja.2021. 100002.

- [38] Z. Zhou, L. Feng, S. Zhang, et al., "The operational performance of "net zero energy building": A study in China," Applied Energy, vol. 177, pp. 716–728, 2016, ISSN: 03062619. DOI: 10.1016/j.apenergy.2016.05.093.
- [39] EQUA Simulation Technology Group, "IDA Indoor Climate and Energy," no. 2014/09/08, 2014. [Online]. Available: http://www.equa-solutions. co.uk/de/software/idaice.
- [40] S. Norge, "SN/TS 3031:2016, Bygningers energiytelse Beregning av energibehov og energiforsyning," 2016.
- [41] Nordpool. "Day-ahead prices." (), [Online]. Available: https://www.nordpoolgroup. com/en/Market-data1/Dayahead/Area-Prices/ALL1/Hourly/?view=table (visited on 05/10/2022).
- [42] NVE. "Spørsmål og svar om strømstøtte for høye priser." (), [Online]. Available: https://www.nve.no/reguleringsmyndigheten/kunde/strom/ spoersmaal-og-svar-om-stroemstoette-for-hoeye-stroempriser/ (visited on 05/10/2022).
- [43] S. Norge, "Ergonomi i termisk milj Analytisk bestemmelse og tolkning av termisk velbefinnende ved kalkulering av PMV- og PPD-indeks og lokal termisk komfort," 2005.

A Appendix 1

Python script for finding the yearly peak load.

```
import numpy as np
⊖import pandas as pd
load = pd.read_csv("/Users/ashildryen/PycharmProjects/pythonProject/S8-18PV.csv", sep=";", names=['time', 'value'])
X_f = load['time'] #pandas.core.series.Series
Y_f = load['value']
x_list_f = load['time'].tolist()
y_list_f = load['value'].tolist()
y_f = np.array(Y_f)
x_f = np.array(X_f)
a=0
i=0
    y = float(y_f[i])
    x = float(x_f[i])
    current = y
    if current > a:
        a = current
```

B Appendix 2

Python script for finding the monthly peak load.

```
import numpy as np
import pandas as pd
load = pd.read_csv("/Users/ashildryen/PycharmProjects/pythonProject/S12.csv", sep=";", names=['time', 'value'])
X_f = load['time'] #pandas.core.series.Series
Y_f = load['value']
x_list_f = load['time'].tolist()
y_list_f = load['value'].tolist()
y_f = np.array(Y_f)
x_f = np.array(X_f)
   y = float(y_f[i])
   x = float(x_f[i])
   current = y
   if current > a:
       a = current
       b = x
print(a)
print(b)
b=0
   i += 1
   y = float(y_f[i])
   x = float(x_f[i])
   current = y
```

34 Ę	↓ if current > a:
	a = current
	b = x
	print("Feb")
	print(a)
	print(b)
	a≞θ
	b≘0
	while i<2160:
	i += 1
	<pre>y = float(y_f[i])</pre>
	x = float(x_f[i])
	current = y
	if current > a:
	a = current
	b = x
	print("Mar")
	print(a)
	print(b)
	a,=0
	b≝0
	owhile i≲2880:
	i += 1
	y = float(y_f[i])
	x = float(x_f[i])
	current = y
	if current > a:
	a = current
	print("Apr")
	print(a)
	print(b)
67	

4.9	Lubile i/3424.
40	i 1= 1
79	y = float(y f[i])
71	$\mathbf{x} = float(\mathbf{x} = f[i])$
72	
73	
76	
75	
76	print("Mai")
77	print(a)
78	print(b)
79	a=0
80	b=0
81	ewhile i<4344:
82	i += 1
83	<pre>y = float(y_f[i])</pre>
84	<pre>x = float(x_f[i])</pre>
85	current = y
86	if current > a:
87	a = current
88	$\dot{\varphi}$ b = x
89	print("Jun")
90	print(a)
91	print(b)
92	a≘θ
93	b≘0
94	∲while i≾5088:
95	i += 1
96	y = float(y_f[i])
97	x = float(x_f[i])
98	current = y
99	<pre>if current > a:</pre>

100	a = current
101 6	b = x
102	
103	print("Jul")
104	print(a)
105	print(b)
106	a≘0
107	b≘0
108 (while i<5832:
109	i += 1
110	y = float(y_f[i])
111	<pre>x = float(x_f[i])</pre>
112	current = y
113 (if current > a:
114	a = current
115 ($b = \mathbf{x}$
116	print("Aug")
117	print(a)
118	print(b)
119	a≘0
120	b≘0
121 (While i<6552:
122	i += 1
123	<pre>y = float(y_f[i])</pre>
124	<pre>x = float(x_f[i])</pre>
125	current = y
126 (<pre>if current > a:</pre>
127	a = current
128 6	
129	print("Sep")
130	print(a)
131	
132 133	

177	
175	i += 1
190	1 = 1
130	$y = float(y_f[1])$
137	x = float(x_f[1])
138	current = y
139	<pre>uf current > a:</pre>
140	a = current
141	b = x
142	print("Okt")
143	print(a)
144	print(b)
145	a≞0
146	b≓0
147	¢while i≤8016:
148	i += 1
149	<pre>y = float(y_f[i])</pre>
150	<pre>x = float(x_f[i])</pre>
151	current = y
152	if current > a:
153	a = current
154	• b = x
155	print("Nov")
156	print(a)
157	print(b)
158	a≞0
159	b≞0
160	e while i≤8760:
161	i += 1
162	<pre>y = float(y_f[i])</pre>
163	<pre>x = float(x_f[i])</pre>
164	current = y
165	if current > a:

C Appendix 3

The 20 first schedules established for regulation of the GSHP.




Appendix 4 D





E Appendix 5

The 15 schedules established during the sensitivity analysis of Case 2.





F Appendix 6

The 15 schedules established during the sensitivity analysis of Case 3.





G Appendix 7

The following pages shows a scientific paper based on some of the results found in the master's thesis.

Assessing energy flexible operation of a residential Zero Emission Building (casaZero)

Åshild Ryen Vespestad, Vojislav Novakovic, Karen Byskov Lindberg

Norwegian University of Science and Technology (NTNU), June 2022

Abstract

With an increasing amount of new and retrofitted buildings qualifying as ZEBs, the interaction between buildings and the grid has increased. Prosuming buildings that both produces and consumes electricity, are likely to both import and export electricity to the grid. Hence, the need to utilized smart controlling of the electricity consumption in buildings is reinforced as an attempt to reduce the stress on the grid. This study is a contribution to increase the knowledge on flexible operation of ZEBs. By implementing 35 different operational schedules for the GSHP of a residential ZEB, the impact on the performance of the building was studied. The KPI analysed in this study was the potential of peak load reduction. It was found to be 50%, when the GSHP was regulated to half the original potential at all time interrupted by two off-periods during the day. It was also found that by reducing the yearly peak load, values for both self-consumption and cost were affected.

1 Introduction

1.1 Background/Motivation

As the building sector today causes one third of the global energy use [1], ZEBs have become an important contribution to reducing the world wide energy use. The European Union goal of reducing the CO2 emission associated with the building sector by 80-95% compared to the 1990 level by 2050 [2], requires reduction in building energy use as well as increased utilization of on-site generated renewable energy.

All buildings require a minimum amount of electricity. Hence, ZEBs depend on on-site renewable electricity production to fulfill the ZEB requirement. Often the renewable electricity sources available are fluctuating, leaving the building dependent on electricity from the grid in hours when the local electricity sources are not supplying sufficient amount of electricity to the building [3]. When the electricity sources produce more electricity than consumed by the building, the excess electricity can be supplied to the grid. This way of buying and selling electricity to the grid characterizes the building as a prosumer [4]. With an increasing number of prosumers intermittently supplying electricity to the grid, the grid companies are facing challenges handling the unpredictability in the grid. Both considering hourly mismatch and seasonal mismatch [5]. This calls for an increased utilization of energy flexibility measures in ZEBs to increase the self-consumption and reduce the peak loads in the building to reduce the pressure on the grid. By the end of 2022 a new law will be implemented in the Norwegian electricity regulations [6]. The new electricity prices includes a variable grid fee based on the monthly peak load. Peak load reduction can therefore be motivated by financial gain as well as decreasing the pressure on the grid.

Since the term ZEB was introduced in 2006 [7], several research papers have been written focusing on optimizing ZEBs. New and improved technology has emerged, enabling communication between technical equipment in the building. This increases the possibility of operating buildings in an optimal way considering cost, CO2 emission, indoor environment etc. Despite this, to the authors knowledge, the main research area on ZEBs has been on optimizing the design rather than the operation. To find out how optimal operation of a ZEB can effect the energy use and CO2 emission, more research on optimization of the operation of a ZEB is coveted.

1.2 Objectives of this research

The objective of this study was to asses energy flexible operation of a detached residential ZEB in Norway, named casaZero. The building was equipped with a GSHP, a water borne space heating system, a balanced ventilation system and PV panels for electricity production. Different operational strategies for the GSHP were tested to find the impact on the performance of the building, focusing on the potential of peak load reduction.

2 Methodology

2.1 casaZero

The building analysed in this study was a ZEB located in Kolsås, close to Oslo. The residential building had a total heated area of 233.6 m². It consisted of basement, 1.floor and 2.floor. The building was equipped with a balanced ventilation system and was heated through water pipes built into the floor. Only the 1. floor and the basement were equipped with the water borne heating system. The water was heated by a 6kW GSHP with an average COP of 4, and stored in a 224L DHW tank. The tank provided hot water for both the space heating system and the DHW tapping system. The south side of the roof was covered with 50 m² of PV panels generating an accurate amount of 10500 kWh yearly. The model of casaZero established in IDA ICE utilized schedules for lighting and equipment found in SN/TS 3031:2016 [8]. The simulation period was set to one year, utilizing a climate file from Oslo, Gardemoen. A detailed description of the building and the model established in IDA ICE is found in [9].

2.2 Key performance indicator

The KPI analysed in this study was the peak load. The aim was to reduce the electricity import peak. In addition to the KPI analysed, two criteria were given for the simulations. The first criteria was made as a result of the ZEB definition,

considering the energy balance. As shown in Equation 1, the total electricity consumption could not exceed the total electricity production for the one year period. For each simulation this was checked to ensure a positive energy balance.

$$El^{imp} < El^{exp} \tag{1}$$

 El^{imp} and El^{exp} is respectively total electricity imported and total electricity exported during a period of one year.

The second criteria was the indoor thermal comfort. It was evaluated by using the percentage of total occupancy hours with thermal dissatisfaction, referred to as the percentage of people dissatisfied (PPD). This was done by separately summing up the total occupancy hours for all zones and the total hours of thermal dissatisfaction. The two numbers were then implemented in Equation 2.

$$PPD = \frac{h^{PD}}{h^{occ}} \tag{2}$$

Where h^{PD} is the hours of people dissatisfied and h^{occ} is the occupancy hours. Only schedules with PPD<10% were qualified as satisfactory based on the standards given by NS-EN ISO 7730:2005 [10].

2.2.1 Peak load

The equation used for calculating the total imported electricity for each time step is shown in Equation 3. A script was established in Python to find the highest value of imported electricity during a year, as well as a script for finding the monthly peak load.

$$El_t^{imp} = El_t^{heat} + El_t^{eq} + El_t^{hvac} + El_t^{light} - El_t^{prod}$$
(3)

 El_t^{imp} is the imported electricity, El_t^{heat} is the electricity consumed by electric heating, El_t^{eq} is the electricity consumed by equipment, El_t^{hvac} is the electricity consumed by the HVAC system, El_t^{light} is the electricity consumed by lighting and El_t^{prod} is the produced electricity, with t representing the time step of the simulation period.

2.3 Base case

In the Base case model the GSHP was regulated by a thermometer in the DHW tank. The GSHP was turned on if the temperature was too low. An electric boiler acted as the top heater, supplying heat to the DHW tank when the GSHP could not supply enough heat.

2.4 Case with GSHP schedules

To evaluate the influence of applying different operational strategies to the GSHP, 20 schedules were established to control the work of the GSHP. The schedules were

named S1, S2, ..., S20. The electric top heating of the DHW was removed, but the thermometer regulating the GSHP was kept. Hence, the GSHP was only operating in hours when both the schedule and the thermometer allowed it.

Each schedule was analysed according to the KPI. The three schedules with lowest result for yearly peak load were selected and a sensitivity analysis was performed on each of the three schedules. The sensitivity analysis was performed by developing 5 new schedules for each of the three chosen schedules. The new schedules had only minor adjustments compared to their origin schedule. A total of 15 new schedules were established during the sensitivity analysis.

3 Results and analysis

3.1 Base case

The following main results were found from the simulation of Base case:

- Total produced electricity: 10 479 kWh
- Total consumed electricity: 9559 kWh
- Total purchased electricity: 5732 kWh
- Total exported electricity: 6653 kWh
- Self-sufficiency: 0.3791
- Peak load: 7788 W
- Cost: 9924.3 NOK

3.2 Case with GSHP schedules

The 15 schedules established during the sensitivity analysis is presented in Figure 1. The row written in red shows the schedules with the lowest yearly peak load without compromising the thermal comfort. The three schedules marked in green are the schedules with best results from the first round of simulation.

Schedule	SC [-]	Peak load [W]	Cost [NOK]	PPD [%]
S7(1)	0,37692517	4145	9790,5	9,81
S7(2)	0,36270583	4411	9734,4	9,77
S7(3)	0,353194	4355	9769	9,75
S7(4)	0,38211529	4627	9660,7	9,8
S7(5)	0,32603356	4842	9766,2	9,78
S9(1)	0,34732003	4798	9654,8	9,76
S9(2)	0,30728642	3896	9763,8	9,76
S9(3)	0,35052978	4885	9673,5	9,74
S9(4)	0,32448456	4901	9750,9	9,71
S9(5)	0,34368623	4510	9793,2	9,76
S20(1)	0,37670432	4238	9808,4	9,79
S20(2)	0,34504328	5021	9470,8	9,71
S20(3)	0,37449012	5380	9405	10,09
S20(4)	0,40187575	5447	9349,9	10,19
S20(5)	0,34074366	4989	9398,9	9,77
S7	0,32640015	4700	9712,3	9,72
S9	0,32886731	4898	9643,3	9,74
S20	0,34216941	4949	9490,5	9,78

Figure 1: An overview of the main results from the sensitivity analysis.

The results in the figure shows that by improving only peak load, the value for both self-consumption and cost are affected. Based on the results in Figure 1 schedule S9(2) gives the lowest yearly peak load. The S9(2) schedule is shown in Figure 2.



Figure 2: GSHP schedule for S9(2).

Figure 3 gives an overview of the peak load found for each month in the Base case compared to S7, S9, S7(1) and S9(2). Though the S9(2) schedule has the lowest yearly peak load, the figure shows that S7(1) puts a lower pressure on the grid for larger parts of the year. For April, the peak load for S7(1) is reduced by 37.4% compared to S9(2) and the average peak load reduction for the 12 months is 11.3%. It can also be noted from the figure that the Base case has a low peak load for the months from April to October. However, the peak loads during the winter months are about doubled compared to the values for S7(1) and S9(2).

In addition to affecting the pressure on the grid, it is reasonable to assume that the difference in peak load for S9(2) and S7(1) will influence the cost. In this study, a cost analysis including the new electricity fee was not performed, however the variety in monthly peak load between the two schedules indicates a lower yearly cost for S7(1).



Figure 3: Monthly peak load for Base case, S7, S9, S7(1) and S9(2).

A load profile for Base case and S9(2) is displayed in Figure 4. The load profile shows the three days during the winter season with the highest total electricity consumption. Though the peaks are mostly coinciding, the peak values are strongly reduced for the S9(2) schedule. The lower and wider peaks found in S9(2) is due to the regulation of the GSHP found in Figure 2. As S9(2) never allows the GSHP to operate with full capacity, the load is distributed on a larger interval to cover the same heating demand as found for the peaks in Base case.



Figure 4: Load profile for the 72 hour periods with highest electricity consumption for winter season for S9(2) and Base case.

The S9(2) schedule turns off the GSHP during periods of the day. Due to this, more peaks are found in the graph for S9(2) compared to base case. This is also seen in Figure 5 where Base case resulted in a smooth curve with little slope, while S9(2) has several minor peaks during the day. During summer, the peaks are slightly higher for S9(2) compared to Base case. However, the off periods of the GSHP during PV production hours causes a decrease in the self-consumption for S9(2).

Though the S7(1) schedule was the best solution considering monthly peak loads, the S9(2) schedule had the highest reduction in yearly peak load with a reduction of 50%. The equivalent number for S7(1) was 47%. Hence, S9(2) was the best schedule found in this study for yearly peak load reduction.



Figure 5: Load profile for the 72 hour periods with highest electricity consumption for summer season for S9(2) and Base case.

4 Conclusion

This study investigated 35 different operational strategies for the GSHP in a residential ZEB. The aim was to find the influence on the energy performance of the building. According to the chosen KPI of the study, S9(2) had the best result. The S9(2) schedules improved the yearly peak load by 50% compared to Base case. However, for 10 out of 12 months the monthly peak loads were higher than for the S7(1) schedule. It was also found in the results of this study that when improving the peak load both self-consumption and cost were affected.

4.1 Further work

Based on the results of this study it was found that further research on the cost effect of the monthly peak load fee soon to be implemented in Norwegian electricity

prices is of interest. As the load curve for winter differed significantly from the load curve for summer, more research on seasonal differences and its influence on different KPIs would be of interest in regions with significant seasonal differences. Furthermore, utilization of optimization tools like GenOpt or AutoMOO would be desired, as the results would give more exact values for the KPI compared to the once found in this study.

5 Acknowledgment

Acknowledgement should be given to NTNU for providing the opportunity to perform this study.

References

- IEA. "Buildings A source of enormous untapped efficiency potential." (), [Online]. Available: https://www.iea.org/topics/buildings (visited on 05/04/2022).
- [2] EU, "Directive (EU) 2018/844 of the European Parlament and of the Council of 30 May 2018," Official Journal of the European Union, no. June 2028, 2018, ISSN: 16113349. DOI: 10.1007/3-540-47891-4{_}10.
- [3] M. G. Fikru, J. Atherton, and C. I. Canfield, "Cost-reflective dynamic electricity pricing for prosumers," vol. 35, 2022.
- Y. Parag and B. K. Sovacool, "Electricity market design for the prosumer era," *Nature Energy*, vol. 1, no. 4, 2016, ISSN: 2058-7546. DOI: 10.1038/nenergy. 2016.32.
- [5] K. B. Lindberg, D. Fischer, G. Doorman, M. Korpås, and I. Sartori, "Costoptimal energy system design in Zero Energy Buildings with resulting grid impact: A case study of a German multi-family house," *Energy and Buildings*, vol. 127, pp. 830–845, 2016, ISSN: 03787788. DOI: 10.1016/j.enbuild.2016. 05.063.
- [6] Lovdata. "Forskrift om endring i forskrift om økonomisk og teknisk rapportering, inntektsramme for nettvirksomheten og tariffer." (), [Online]. Available: https://lovdata.no/dokument/LTI/forskrift/2021-06-10-1904 (visited on 04/27/2022).
- [7] C. Ionescu, T. Baracu, G. E. Vlad, H. Necula, and A. Badea, "The historical evolution of the energy efficient buildings," *Renewable and Sustainable Energy Reviews*, vol. 49, pp. 243–253, 2015, ISSN: 18790690. DOI: 10.1016/j.rser. 2015.04.062.
- [8] S. Norge, "SN/TS 3031:2016, Bygningers energiytelse Beregning av energibehov og energiforsyning," 2016.

- [9] Å. R. Vespestad, "Assessing energy flexible operation of a residential Zero Emission Building (casaZero)," 2022.
- [10] S. Norge, "Ergonomi i termisk milj Analytisk bestemmelse og tolkning av termisk velbefinnende ved kalkulering av PMV- og PPD-indeks og lokal termisk komfort," 2005.



